

No. 2008–10

HOW PRODUCTIVE IS PUBLIC CAPITAL? A META-ANALYSIS

By Pedro R.D. Bom, Jenny E. Ligthart

January 2008

ISSN 0924-7815



How Productive is Public Capital? A Meta-Analysis^{*}

Pedro R.D. Bom[†] Tilburg University

Jenny E. Ligthart[‡] Tilburg University, University of Groningen, and CESifo

January 18, 2008

Abstract

The paper analyzes the contribution of public capital to private output using several metaanalytical techniques. Both fixed and random effects models are estimated by Weighted Least Squares. Sample overlap across studies is explicitly controlled for by employing a 'full' Generalized Least Squares estimator. The weighted average output elasticity of public capital amounts to 0.08 after correcting for publication bias. A substantial part of the heterogeneity across studies is explained by study design parameters, such as econometric specification, estimation technique, empirical model, type of public capital, and level of aggregation of public capital data. The large elasticities of public capital found in the early literature seem to be caused by either unidentified (but present) cointegrating relationships or spurious relationships in national time series.

JEL codes: H540

Keywords: public capital, infrastructure, public investment, meta-analysis, meta-regression analysis, publication bias

^{*}The authors would like to thank Jan-Egbert Sturm for helpful comments. Jenny Ligthart gratefully acknowledges financial support from the Dutch Ministry of Finance.

[†]CentER and Department of Economics, Tilburg University, P.O. Box 90153, 5000 LE Tilburg, The Netherlands. Phone: +31-13-466-2068, Fax: +31-13-466-4032, E-mail: p.r.duartebom@uvt.nl.

[‡]Corresponding Author: CentER and Department of Economics, Tilburg University, P.O. Box 90153, 5000 LE Tilburg, The Netherlands. Phone: +31-13-466-8755, Fax: +31-13-466-4032, E-mail: j.ligthart@uvt.nl.

1 Introduction

Discussions among academics and policy makers about the contribution of the public capital stock to private output growth have been ongoing during the last two decades. Recently, these debates have revived within the European Union (EU), following the renewed interest in fiscal policy rules—in the form of ceilings on the deficit-to-GDP and public debt-to-GDP ratios—and their potential negative impact on public capital formation. Indeed, in many instances, policy makers find it easier to cut back on infrastructure investment rather than on current expenditure. To provide input to the public capital debate, it is of importance to have insight into the stylized facts on the linkage between private output and public capital formation.

Many authors have tried to determine the contribution of public capital to private output by estimating a production function that includes the public capital stock as an input (the so-called production function approach). Aschauer (1989a, 1989b, 1990) was one of the first to investigate this issue for the United States in an attempt to explain the productivity growth slowdown in the 1970s.¹ Indeed, in the United States and various OECD countries, investments in the public capital stock fell and aggregate labor productivity growth declined slightly later. Aschauer (1989a) found that a 1 percent increase in the public capital stock increased private output by 0.39 percent, suggesting that public capital is an important determinant of output. Since then, many studies have been undertaken for the United States and various other OECD countries. More recently, attention has also been focused on the productivity effects of public capital in developing countries (e.g., Ram, 1996). The findings of these studies generally range from no significant effect to a strongly positive effect of public capital on output. Some studies, though not that many, even find significantly negative results. So far, researchers have not attached much priority to reconciling these differences.

Although various authors have reviewed the literature on the productivity of public cap-

¹Mera (1973) was the first to estimate a production function including some form of public capital, which he refers to as 'social capital,' for nine Japanese regions. This work was followed by work of Ratner (1983) and Da Costa, Ellson, and Martin (1987).

ital,² none of them has applied a systematic meta-analysis yet.³ The aim of the paper is to fill this gap. Drawing on Stanley and Jarrell (1989) and Stanley (2001), meta-analysis can be defined as a body of statistical methods to summarize, evaluate, and analyze empirical results across studies. A problem with conventional reviews of the literature is that studies are difficult to compare, owing to differences in the empirical model, econometric specification, estimation method, and data definitions. Meta-analysis presents a more systematic and objective way to summarize empirical results. It allows us to explain the wide study-to-study variation by the researcher's choices made on research design in the analysis. In this way, a *meta*-output elasticity of public capital can be derived, which researchers and policy makers can use in their analyses without conducting any empirical research themselves.⁴

Our meta-sample pertains to studies employing public capital as an input into production. The sample covers all relevant studies up to and including the year 2006, yielding a meta-data set of 76 studies. Instead of using all available observations, we include a single observation per study, which allows us not only to control for dependency across multiple observations taken from a single study, but also to increase—by focusing on the highest quality estimates only—the accuracy of the true effect estimate. We compute both fixed and random effects estimates of the true underlying output elasticity while controlling for publication bias. In contrast to the work of Stanley (2005), we explicitly test for bidirectional publication bias that is potentially asymmetric.

Besides a simple meta-analysis, we conduct a more complex meta-regression analysis. We test for a large set of potential determinants of heterogeneity across studies. We estimate a fixed and random effects meta-regression model using Weighted Least Squares (WLS). In the fixed effects specification, the equation is multiplied by the inverse of the within-study

 $^{^{2}}$ See the studies by Munnell (1991, 1992), Gramlich (1994), Pfahler et al. (1996), Button (1998), Sturm et al. (1998), Button and Rietveld (2000), Mikelbank and Jackson (2000), IMF (2004), and Romp and De Haan (2007).

 $^{^{3}}$ We are aware of only one study, that of Button (1998), who has made a first (but very incomplete) attempt to explain observed heterogeneity across studies on public capital. Button's (1998) meta-regression analysis covers only 26 studies (published during 1973–1994), which yields a meager total of 28 data points. He finds one significant moderator variable, that is, whether a study pertains to the United States.

⁴Meta-analysis has a long-standing tradition in psychology and medical research. Environmental and transport economists were the first to apply meta-analysis in economics in the 1980s. Since then, it has been picked up by other fields in economics such as labor economics (e.g., Card and Krueger, 1995), industrial organization (e.g., Button and Weyman-Jones, 1992), and international economics (e.g., De Mooij and Ederveen, 2003; and Rose and Stanley, 2005).

standard deviation. In the random effects specification, however, we use the sum of the within and between variation as weights. We control for dependency of estimates across countries by including country-fixed effects and correct for heteroscedasticity by employing White standard errors and also clustering of standard errors.

Our methodological innovation is that we account for the degree of sample overlap in a random effects model by using a 'full' Generalized Least Squares (GLS) procedure. Note that the WLS procedure—which is a stripped down version of GLS, which we call 'partial' GLS—leaves unused information on the error-covariances of the original estimates. Because various authors make use of identical or very similar samples, thus creating sample-dependency across estimates, the off-diagonal elements of the variance-covariance matrix are non-zero. We employ a simple procedure to calculate the degree of sample overlap, which is subsequently used to proxy error correlation. The error variances, in turn, are derived from the standard errors of the original estimates. By weighting the original measurements by the variance-covariance matrix obtained in this manner, a full GLS estimator is obtained. We show that a substantial amount of heterogeneity across studies is explained by study design characteristics such as the econometric specification, estimation technique, empirical model, type of public capital used, and level of aggregation of public capital data.

The remainder of the paper is structured as follows. Section 2 discusses definitions, presents various approaches used to estimate the output elasticity of public capital and discusses empirical results. In addition, it gives an overview of the criticisms launched against the most widely used methodology, that is, the production function approach. Section 3 describes the meta-sample and presents the meta-analysis results. Section 4 discusses and estimates publication bias and applies a publication bias correction to our meta-analysis. Section 5 sets out the meta-regression model and discusses the meta-regression results. Section 6 concludes.

2 Public Capital and Private Output

How do we define and measure the public capital stock? Which approaches exist to measure the productivity of public capital? What size of the output elasticity of public capital do empirical studies typically find? This section addresses these important questions before it ventures into a review of the methodological issues.

2.1 Defining Public Capital and Output

Gramlich (1994, p. 1177) defines *infrastructure capital* from an economic point of view as 'large capital intensive natural monopolies such as highways, other transportation facilities, water and sewer lines, and communications systems.' Although most of these systems are publicly owned, in some cases they are privately owned. For example, a firm that constructs its own road to connect itself to the main highway. The literature generally defines infrastructure capital based on ownership. It is the public component of infrastructure capital that most people have in mind when they talk about public capital. The issue of definitions is more subtle, however.

Most empirical studies employ a 'narrow definition' of public capital that includes the tangible capital stock owned by the public sector excluding military structures and equipment. More specifically, it consists of core infrastructure (i.e., roads, railways, airports, and utilities such as sewerage and water facilities), hospitals, educational buildings, and other public buildings. Some authors use a 'broad definition' of public capital by also including human capital investment (e.g., Garcia-Milà and McGuire, 1992), or health and welfare facilities (e.g., Mera, 1973). The latter components are hard to measure, explaining why most authors focus on the narrow concept of public capital.

To arrive at an estimate of the stock of public capital at a particular moment in time, researchers determine an initial value of the capital stock to which they add gross investment flows and subtract technical depreciation of the existing capital stock (based on the expected life spans of its components).⁵ The majority of studies employ public capital stocks defined at the national level including *all* levels of government (e.g., Aschauer, 1989a), whereas others deal with capital stocks estimated for regions (e.g., Garcia-Milà and McGuire, 1992). Some studies only consider capital that is owned by local governments (e.g., Evans and Karras, 1994a), which does not take into account regionally installed capital owned by the

⁵See Sturm and De Haan (1995) for further details on this so-called perpetual inventory method.

central government. We are aware of only a few studies estimating capital stocks at the city/metropolitan level (e.g., Duffy-Deno and Eberts, 1991; and Kemmerling and Stephan, 2002).

The measure of output—which is used as the dependent variable in the econometric analysis, see Section 2.3—varies across studies. Most studies use either real *net* output of the private sector (e.g., Ratner, 1983)⁶ or real Gross Domestic Product (GDP)—or, alternatively, real Gross State Product (GSP), when the data is at the state level for the United States exclusive of public sector output (e.g., Finn, 1993). Because government output is typically not exchanged on markets, it is hard to measure. In the National Accounts, it is equated to the wage bill of the public sector. Although we are primarily interested in measuring the contribution of public capital to private output rather than total output, not every study employs a measure of output that corrects for public sector production. The latter is typically the case of studies using data for emerging markets or developing countries, where the only available measure of output is total GDP (e.g., Ram, 1996).

2.2 Empirical Methodologies

The literature has distinguished various approaches to study empirically the link between private output and public capital. The production function approach, which is the most widely known and applied, considers the stock of public capital either as a separate input in private production (which we call the 'pure production function approach') or as a factor affecting multifactor productivity (which is known as the growth accounting approach; see Hulten and Schwab, 1991b). In both cases, public capital is assumed to be strictly exogenous. Evidently, because the growth accounting approach does not yield direct estimates of the output elasticity of public capital, it will not be covered in our empirical analysis. Closely linked to the production function approach is the production frontier approach, which departs from the former by taking into account that public capital may increase potential production without necessarily increasing actual production (e.g., Delorme et al., 1999). In an efficient steady state, the production frontier approach is equivalent to the standard production function

 $^{^{6}}Net$ output (which equals gross value-added) is obtained by subtracting the value of intermediate goods and services from *gross* output of the private sector.

approach. We take this distinction into account in our empirical analysis.

Another methodology is the Vector Autoregression (VAR) approach, which analyzes the relationships between public capital, private inputs, and private output without imposing a theoretical structure. The multi-equation VAR approach models every endogenous variable as a function of its own lagged value and the lagged values of the other endogenous variables and thus can assess whether there is any feedback from private sector variables to the public capital stock. We do not include pure VAR studies in our analysis because in this framework it is hard to disentangle the direct effect (i.e., the production elasticity) from the feedback effects. Some authors (e.g., Ligthart, 2002), however, employ Johansen's (1988) method—which makes use of the VAR technique—to check whether the variables in the production function are cointegrated (see Section 2.5). The latter studies will be included in our sample.

Other approaches are the following two. The cross-country growth regressions approach, which specifies a reduced-form equation to estimate—using cross-section or panel data—the relationship between per capita private output growth and the public investment-to-GDP ratio. The growth regressions approach should be distinguished from those studies explicitly embedding a production function in the framework (which we will call the 'production function-based approach'). We will classify the latter under the production function approach if an output elasticity of public capital is or can be derived. Last but not least, the behavioral approach—coined as such by Sturm et al. (1998)—which employs cost or profit functions to assess whether public capital reduces firms' production costs or increases firms' profits. This last approach does not specify a direct relationship between public capital and private output, so it will not be covered in the empirical section.

2.3 The Production Function Approach

The corner stone of the production function approach is a technological relationship that incorporates the stock of public capital at time t, denoted by G_t , as an input:

$$Y_t = A_t F\left[K_t, L_t, G_t\right],\tag{1}$$

where Y_t is real aggregate private output within some area (region or country), A_t is an index of economy-wide productivity, K_t denotes the stock of (non-residential) private fixed capital, and L_t denotes employment (typically measured by total hours worked). Equation (1) shows that public capital may affect aggregate real private output in two ways. First, a direct effect, that is, $\partial Y_t / \partial G_t > 0$. The idea is that the services of public capital are proportional to the stock of public capital—which is generally assumed to be a *pure* public good—and contribute to production in that way. Second, public capital may raise the marginal productivity of private factors of production, that is, $\partial^2 Y_t / (\partial K_t \partial G_t) > 0$ and $\partial^2 Y_t / (\partial L_t \partial G_t) > 0$.

Most studies employ a Cobb-Douglas production function:⁷

$$Y_t = A_t K_t^{\alpha} L_t^{\beta} G_t^{\theta}, \qquad \alpha, \beta, \theta > 0, \tag{2}$$

where $\theta \equiv \partial \ln Y_t / \partial \ln G_t$ is the output elasticity of public capital, which is hypothesized to be positive. This specification models public capital and private inputs as cooperative factors of production. By taking natural logarithms on both sides of (2), we get a linear relationship:

$$\ln Y_t = \ln A_t + \alpha \ln K_t + \beta \ln L_t + \theta \ln G_t.$$
(3)

Equation (3) can readily be estimated in logarithmic (log) levels or first differences of log levels (i.e., growth rates) to arrive at estimates of α, β , and θ . As can be seen from (3), the productivity index enters the equation in an additive way. Following Ratner (1983), many studies include a constant and a time trend as a proxy for technological progress (i.e., $\ln A_t = a_0 + a_1 t$, where $a_0, a_1 > 0$).

Incorporating public capital into the production function raises the issue of returns to scale in production. Imposing the restriction of constant returns to scale (CRTS) across all inputs, which is represented by $\alpha + \beta + \theta = 1$, yields the specification employed by the majority of studies:

$$\ln(Y_t/K_t) = \ln A_t + \beta \ln(L_t/K_t) + \theta \ln(G_t/K_t).$$
(4)

Equation (4) features decreasing returns with respect to private inputs taken together.⁸ Instead of using private capital productivity, $\ln(Y_t/K_t)$, some studies subtract $\ln L_t$ from both

⁷Some studies use the more general translog production function, which includes also quadratic and interaction terms for each input. Early adopters of the translog specification are, amongst others, Merriman (1990), Pinnoi (1994), and Damalgas (1995).

⁸An alternative model assumes CRTS in both private inputs (represented by $\alpha + \beta = 1$; see Mas et al., 1994), allowing for increasing returns to scale across all inputs (i.e., $\alpha + \beta + \theta > 1$).

sides of (3) and impose CRTS so as to arrive at the logarithm of labor productivity as the dependent variable.

2.4 Some Empirical Results

The output elasticity of public capital can be used to derive the marginal productivity of public capital, that is, $\partial Y_t/\partial G_t = \theta(Y_t/G_t)$, which equals the effective rate of return on public capital.⁹ To assess whether investments in public capital are worthwhile, policy makers generally compare the marginal productivity of public capital with the marginal productivity of private capital, which equals the real rate of interest in a competitive market.

Gramlich (1994) argues that the rate of return on public capital derived by Aschauer (1989a) is too large to be credible. Indeed, depending on the year of measurement of $(Y/G)_t$, returns vary between 60 to 80 percent. The marginal output gain of an additional unit of *private* capital estimated from Aschauer's equation amounts to 30 percent, suggesting a difference between public and private capital of a factor two to three. Aschauer (1990) points to the high rate of return found in R&D studies to justify the large output elasticities found in the early literature. Gramlich (1994) claims, however, that a large share of public capital is directed at less productive sectors of the economy, such as waste and pollution abatement, which is likely to contribute little to national output. There is no reason to be pessimistic about the empirical evidence. First, studies published in the 1990s find more realistic values of the output elasticity of public capital. Second, most studies find a positive and statistically significant output elasticity of public capital. Indeed, Ligthart (2002) derives an unweighted average of the output elasticity of public capital of 0.25 for OECD countries if the production function is estimated in logarithmic levels.

The first author studying the output effect of public capital in a regional context is Mera (1973), who analyzes nine Japanese regions, employing a broad definition of public capital. Since then, various authors¹⁰ have found elasticities at the regional level that are much smaller

⁹Here it is assumed that public capital is remunerated based on its marginal productivity. Aaron (1990) argues that in the presence of government pricing inefficiencies and the absence of markets for government services this is not a very realistic assumption.

¹⁰Munnell (1990b), Eisner (1991), Garcia-Milà and McGuire (1992), Evans and Karras (1994a), and Holtz-Eakin (1994).

than those from analyses using aggregated data for a single country. This can be attributed to spillover effects, that is, some of the beneficial effects of public capital accrue to neighboring regions. In a Nash equilibrium, these spillovers are not internalized, so that both regions may end up with a less than socially optimal public capital stock. Spillovers can be formalized as follows:

$$Y_{i,t} = A_{i,t} K_{i,t}^{\alpha} L_{i,t}^{\beta} G_{i,t}^{\theta} G_{j,t}^{\eta}, \qquad \eta > 0,$$
(5)

where G_i is the public capital stock of the home region *i*, G_j is the public capital of the neighboring region *j*, and $\eta > 0$ is the spillover effect. On the significance of the spillover effect, however, no consensus has emerged in the literature yet. Studies by Holtz-Eakin and Schwartz (1995a, 1995b) and Boarnet (1998) find little evidence of spillover effects. Indeed, studies at the aggregate level measure only the *net* effect. Backwash effects, such as congestion and resource exploitation, or displacement effects (i.e., new infrastructure shifts economic activity to other locations) may exceed any positive gross benefits of infrastructure.

The composition of the public capital stock matters for its effect on private production. Core infrastructure is more productive than other types of public capital, such as educational and office buildings and hospitals. Accordingly, empirical studies employing a broad definition of public capital (which necessarily includes less productive components), find a lower θ than studies focusing on core infrastructure (cf. Sturm and De Haan, 1995; and Ligthart, 2002).

2.5 Criticisms of the Production Function Approach

Various authors have criticized Aschauer's model for being misspecified due to the omission of relevant variables. Tatom (1991) makes a case for including energy prices in the production function to account for supply shocks. For example, the rising oil prices of the 1970s may have depressed capital use. Gramlich (1994) criticizes Tatom's approach for mixing production functions and cost functions. Instead of including energy prices, a measure of the quantity of energy use in production should be employed. The study by Vijverberg et al. (1997), for instance, includes imported raw materials in the production function.

Another specification issue concerns the modeling of the effect of the business cycle on factor use. For this purpose, some studies incorporate a capital utilization rate—or, alternatively, the unemployment rate—in the regression equation.¹¹ Because authors use log-linearized empirical models, capacity utilization enters in an additive fashion. Consequently, it affects all factors across the board, which is a restrictive assumption. Some studies, for example, Ratner (1983), have already adjusted the data and thus do not add a separate regressor. The majority of studies, however, do not correct for the business cycle. If one is interested in estimating long-run elasticities of output with respect to factor inputs, then it makes sense to disregard business cycle effects (e.g., Nourzad, 2000; and Ligthart, 2002). Instead, the aim of most studies is to estimate short-run elasticities of production. Not controlling for the business cycle in this case is likely to bias the estimates downwards.

Some of the early studies have been criticized for not properly accounting for common trends. Generally, time series on GDP and the public capital stock contain a unit root or, in other words, they are non-stationary. If variables are non-stationary, the usual test statistics have nonstandard distributions, implying that the application of standard inference procedures gives rise to misleading results. In particular, one may find spurious relationships between outputs and inputs. Some studies have, therefore, proposed to eliminate time trends in variables by taking first differences.¹²

Two criticisms were raised against first differencing. First, the growth rate of private output in a particular year is not strongly correlated with the growth rate in the public capital stock during that same year as lagged effects are likely to be important. Moreover, equations estimated in first differences often yield implausible coefficients for private inputs (see Sturm and De Haan, 1995). Second, by first differencing information on a possible long-run equilibrium relationship between a set of non-stationary time series (in which case variables are cointegrated) may be thrown away. This shifts the focus of the analysis away from the long-run effects of public capital to the short-run effects. Instead of first differencing, the variables should be tested for cointegration. In the mid-1990s, various authors have either employed Engle and Granger's (1987) cointegration test or Johansen's (1988) variant of this test, giving rise to mixed results.

¹¹For example, Aschauer (1989a), Hulten and Schwab (1991a), and Sturm and De Haan (1995) were early adopters of this specification.

 $^{^{12}}$ See, for example, Aaron (1990), Tatom (1991), and Sturm and De Haan (1995).

Equations (1)–(4) assume G_t to be strictly exogenous, implying that causality runs from public capital to private output. Some authors (e.g., Munnell, 1992; and Gramlich, 1994) have pointed to the lack of attention paid to feedback effects. The direction of causality may run from private output to public capital rather than the other way around. Indeed, a higher rate of output growth may generate favorable budgetary conditions (via higher tax receipts), which facilitate an increase in public investment. During the last decade, various authors¹³ have employed VAR models with a view to capturing the dynamic interactions between output, public capital, and private capital.

Some authors solve the endogeneity problem by using a more traditional econometric tool, namely the Instrumental Variables (IV) estimator. The choice of instrumental variables is usually not an easy task, but in a time-series or panel data context lags of the independent variables emerge as natural instruments to be employed. If a positive effect running from output to public capital exists, then Ordinary Least Squares (OLS) estimates of θ in a single equation model like (3) or (4) are known to be exaggerated. Therefore, IV estimates are likely to be smaller than OLS estimates. Baltagi and Pinnoi (1995), for instance, use panel data for the United States to arrive at a pooled OLS estimate of $\theta = 0.16$; the reported IV estimate of θ , however, is only 0.02.

3 Meta-Analysis

This section provides a description of the meta-data set and conducts a simple meta-analysis with a view to assess a meta-output elasticity of public capital.

3.1 The Meta-Data Set

Table A.1 (see Appendix) shows the set of studies reporting estimates of the output elasticity of public capital using (or based on) the production function approach.¹⁴ In total, 76 studies

¹³McMillin and Smith (1994), Otto and Voss (1996), Batina (1998), Flores de Frutos et al. (1998), Pereira and Roca Sagales (1999), Sturm et al. (1999), Ligthart (2002), and Pereira and Roca Sagales (2003) amongst others.

¹⁴Studies dealing with translog production functions were ignored. Converting the estimated parameters to a single output elasticity is not straightforward. Trying to obtain the relevant standard errors is an even more daunting task.

were coded and included in the meta-data set. To obtain a sample of studies representative of the true population, we used a variety of searching methods.¹⁵ We started by checking the references in the overview papers of Sturm et al. (1998) and Romp and De Haan (2007), which together provide a very comprehensive coverage of relevant papers up to 2004.¹⁶ From these sources, we obtained 55 usable references.¹⁷ We then searched for papers citing Aschauer (1989a) in Thomson's *Web of Science*, which allowed us to add eight papers to our meta-data base. We also used the Internet search engine Google *Scholar* and searched for words such as 'public capital' and 'public infrastructure,' each in combination with 'output' or 'productivity,' which yielded another 13 papers (of which six are working papers). Roughly 18 percent (14 out of 76) of the papers are unpublished. Out of 51 published papers, six are published in top-20 journals (based on the ranking of Kodrzycki and Yu (2006)). The data set encompasses single-country studies for 13 different countries and 10 cross-country analyses.

The issue of how many estimates to include in the meta-data set when each study reports more than one is still a controversial issue. Some authors claim that all available estimates (referred to as 'measurements') should be included (e.g., Bijmolt and Pieters, 2001), whereas others are strong believers of selecting only one measurement for each study (e.g., Stanley, 1998; and Van der Sluis et al., 2005). Including all measurements from each study raises two problems. First, it creates dependency among measurements taken from a single study, which we call 'measurement dependency.' Note that dependency across studies may also be present because some studies rely on identical or very similar data sets (which we refer to as 'sample overlap,' see Section 5). Second, studies with a large number of measurements would receive a disproportionate weight in the sample, giving rise to sampling bias (cf. Stanley, 2001). In our sample, the total number of data points is 706, yielding an average number of 9.3 per study. However, the distribution of available data points across studies is highly skewed. The four papers with the largest number of estimates account for 112 estimates (16 percent of total), the first 10 papers report 243 estimates (34 percent) and the top half of the ranked sample

¹⁵See White (1994) for a review of the general procedures for searching and retrieving papers.

¹⁶In addition, we also checked the overview papers by Pfahler et al. (1996), Button (1998), Mikelbank and Jackson (2000), and IMF (2004).

¹⁷The initial data base of studies was much larger. Not all studies could be included owing to missing standard errors, which are a key input into the analysis.

yields 623 estimates (88 percent). Aschauer's (1989a) study reports the largest number of estimates (36 in total), thus receiving a weight six times larger than studies reporting only one estimate (of which only six are included in the sample).

Aside from statistical considerations, there is a more fundamental reason why we include only one measurement per study, that is, we want to measure 'the' true output elasticity of public capital (either measured in a fixed or random effects context). To uncover the value of this parameter as accurately as possible, only those measurements that come closest to the true effect should be used. In each study no more than one measurement can reasonably meet this criterion. Often, the authors themselves consider many of their estimates senseless, which can therefore be discarded upfront.¹⁸

To address these issues, we include only one measurement for each study, which raises the issue of selecting a measurement from multiple measurements. In a few cases, the authors come up with what they consider their 'preferred estimate.' More often than not, however, the choice of the measurement is not clear-cut. In such cases, we apply a set of predefined selection rules. We let consistency prevail over efficiency and pick the estimate that results from the most sophisticated econometric method (cf. Stanley, 1998). For instance, IV is preferred over OLS estimation and panel fixed effects are considered to be superior to pooled OLS and panel random effects. When the disaggregation of total public capital in subcategories proves significant, we select the broadest category. Finally, we choose the estimate from the most parsimonious model as long as the imposed restrictions are not rejected statistically. Following Stanley (1998), when multiple measurements still remain, we average across them. Consequently, we also need to average any moderator variable we want to include, which makes it harder to interpret the coefficients derived from a meta-regression.¹⁹ In view of this, we use this strategy only in a few cases in which differences in estimates are caused by study characteristics that are not included in our set of moderator variables (see Section 5).

Estimates in our sample vary from -0.175 to 0.917, with an arithmetic (or 'simple')

¹⁸Take panel data studies as an example. For comparison purposes, researchers typically report pooled OLS, random effects, and fixed effects estimates. If the fixed effects model is statistically preferred, as is often the case, then both OLS and random effects estimates are inconsistent. Unless the sole objective of the meta-analysis is to explain the heterogeneity created by the use of different statistical methods, these inconsistent estimates should obviously not be used in a meta-analysis.

¹⁹Some studies (e.g., Rosenthal, 1991) average across all measurements.

average of 0.193 and a standard deviation of 0.198, showing quite some variation. Indeed, we expect a substantial amount of variation given that studies differ along several dimensions. Eight studies find negative estimates, whereas 68 report positive coefficients. We find that the median of 0.159 is smaller than the sample average; thus the distribution is asymmetric, potentially indicating publication bias. Restricting our analysis to 65 papers applying the *pure* production function approach does not change the results much. Only when we exclude papers reporting possibly spurious estimates²⁰ do the mean and median (0.178 and 0.140, respectively) slightly decrease, though total and average variation remain large.

3.2 Pooling Estimates

When pooling estimates to arrive at a 'meta'-estimate, the issue of the degree of homogeneity of the estimates needs to be addressed. Two approaches to deriving a meta-estimate of θ can be distinguished. The first approach is the fixed effects model, which assumes that all studies are estimating a common true effect. More formally, denoting the estimate reported in each study in the meta-sample of size N by $\hat{\theta}_i$, and the unknown population (or 'true') effect that is estimated by θ_i , we can write:

$$\hat{\theta}_i = \theta_i + \mu_i, \qquad \text{for } i = 1, ..., N,$$
(6)

where μ_i is the sampling error satisfying $E(\mu_i|\theta_i) = E(\mu_i) = 0$ (where *E* denotes the expectations operator) if each estimate, $\hat{\theta}_i$, is unbiasedly estimating θ_i . The conditional variance of $\hat{\theta}_i$ is defined as $V(\hat{\theta}_i|\theta_i) = V(\mu_i)$ (which represents the within-study variance), whereas its unconditional counterpart is $V(\hat{\theta}_i) = V(\theta_i) + V(\mu_i)$. If all studies are estimating a common true effect (i.e., $\theta_0 = \theta_i$ for all *i*) then the conditional and unconditional variances of $\hat{\theta}_i$ are equal (i.e., $V(\hat{\theta}_i) = V(\mu_i)$). In other words, all variation is due to sampling error. The random effects model assumes that θ_i is drawn randomly from an $iid(\theta_0, \sigma_{\theta}^2)$ distribution, where σ_{θ}^2 is the between-study variance (reflecting heterogeneity across studies). The unconditional variances of θ_i then becomes: $V(\hat{\theta}_i) = \sigma_{\theta}^2 + V(\mu_i)$. Intuitively, the total variability of estimates across studies is composed of *pure* heterogeneity and sampling error.

 $^{^{20}}$ An estimate is considered to be potentially spurious if an equation is estimated in levels without testing for a cointegrating relationship.

For the fixed effects model it suffices to estimate θ , whereas for the random effects model we also need to estimate σ_{θ}^2 . In both models, a simple unbiased estimator of θ is:

$$\bar{\theta} = \frac{\sum_{i=1}^{N} w_i \hat{\theta}_i}{\sum_{i=1}^{N} w_i},\tag{7}$$

which is a weighted average of the sample estimates, $\hat{\theta}'_i$ s, where the w'_i s are weights. Although $\bar{\theta}$ is an unbiased and consistent estimator of θ for any choice of weights, there is only one estimator minimizing its variance, that is, $w_i = 1/V(\hat{\theta}_i)$ (see Hedges, 1994). Intuitively, more precise estimates receive more weight when averaged. In the fixed effects model, $V(\hat{\theta}_i) = V(\hat{\theta}_i | \theta_i)$, so that the weights are calculated from the standard errors of $\hat{\theta}_i$. In the random effects model, however, the weights are given by $w_i = 1/[\sigma_{\theta}^2 + V(\hat{\theta}_i | \theta_i)]$.²¹ Obviously, the simple mean is just a specific case of (7), where the weights are chosen to be the same and equal to 1/N.

Which model is preferable? On statistical grounds the question reduces to testing whether σ_{θ}^2 is statistically different from zero. If $\sigma_{\theta}^2 = 0$, then estimates differ from each other only due to sampling error, which suggests that a fixed effects model should be preferred. However, if $\sigma_{\theta}^2 > 0$, then a random effects model is called for. Because studies differ along several dimensions (i.e., functional specification, econometric technique, definitions of aggregate output and public capital variables, etcetera), sampling error is unlikely to be the sole factor of variation. Consequently, the random effects model becomes the more plausible candidate in our case. A more formal answer is given by the Q test of homogeneity (cf. Shadish and Haddock, 1994):

$$Q = \sum_{i=1}^{N} w_i \hat{\theta}_i^2 - \frac{\left(\sum_{i=1}^{N} w_i \hat{\theta}_i\right)^2}{\sum_{i=1}^{N} w_i},$$
(8)

which under the null hypothesis of homogeneity is χ^2_{N-1} distributed.

The left panel of Table 1 reports the estimates of θ for both fixed and random effects models and their 95 percent confidence bounds. For the whole meta-sample of 76 reported estimates, the fixed effects estimate of θ is 0.092, less than half of that obtained using a simple average. The true effect is between 0.088 and 0.097 with 95 percent confidence, implying that $\theta = 0$ can be rejected. The Q-test strongly rejects the null hypothesis of homogeneity, suggesting that

²¹Note that $\hat{\sigma}_{\theta}^2 = [Q - (N - 1)]/c$, where $c \equiv \sum_{i=1}^N w_i - (\sum_{i=1}^N w_i^2 / \sum_{i=1}^N w_i)$ and Q is defined in (8). Note that we use $w_i = 1/V(\hat{\theta}_i)$ in the definition of Q.

		4	O T MDTTCM		TITECTION			r untication	n blas Corr	ecrion	
			Confiden	ce Interval				Confidenc	ce Interval ^a		
	N	$\bar{\theta}$	Lower	Upper	Q^{b}	$\hat{\sigma}_{ heta}^2$	$\bar{\theta}^c$	Lower	Upper	Q^b	$\hat{\sigma}^2_{ heta}$
All Studies											
Fixed	26	0.092	0.088	0.097	1161.3	0.007	0.064	0.029	0.099	819.6	0.005
Random	26	0.152	0.129	0.175			0.081	0.053	0.109		
All Studies Excluding VAR											
Fixed	74	0.090	0.086	0.095	1067.4	0.006	0.064	0.029	0.099	753.5	0.004
Random	74	0.147	0.124	0.169			0.075	0.049	0.102		
Pure Production Function											
Fixed	65	0.090	0.085	0.094	1031.4	0.006	0.064	0.028	0.100	740.9	0.004
Random	65	0.151	0.126	0.175			0.080	0.051	0.109		
$Core \ In frastructure$											
Fixed	13	0.128	0.109	0.146	201.9	0.020	0.041	-0.057	0.140	160.3	0.001
Random	13	0.209	0.120	0.297			0.083	-0.046	0.211		
$Regional \ Data$											
Fixed	41	0.076	0.070	0.083	582.8	0.008	0.043	0.023	0.062	399.1	0.002
Random	41	0.110	0.080	0.141			0.066	0.034	0.099		
$Non-Spurious^d$											
Fixed	68	0.106	0.101	0.112	1007.5	0.009	0.076	0.020	0.131	788.2	0.004
Random	68	0.144	0.118	0.170			0.080	0.051	0.110		

Table 1: Fixed and Random Effects Estimates of the Output Elasticity of Public Capital

we should employ the random effects model. In this model, the estimate of θ becomes 0.152, featuring a confidence interval of [0.129, 0.175]. Note that the confidence interval is now much wider, because under random effects the variance of $\bar{\theta}$ reflects not only sampling variation but also the term σ_{θ}^2 . The estimate of θ hardly changes for the restricted sample that includes pure production function studies only. If potentially spurious results are also excluded the random effects estimate is 0.144, which is somewhat smaller than the estimate for either the complete sample or the pure production function. This apparently counterintuitive result is explained by the contribution of the heavily weighted (but small) estimate of 0.059 by Otto and Voss (1996), which is coded as a possibly spurious result.

Table A1 reveals three studies with extremely small values for the standard errors, namely, Garcia-Milà and McGuire (1992), Bajo-Rubio and Sosvilla-Rivero (1993), and Otto and Voss (1996). Consequently, these studies carry a large weight in the meta-analysis. If these measurements were excluded, we would find fixed and random effects estimates of 0.121 and 0.160, respectively. Consequently, the value of $\bar{\theta}$ would rise by not more than 5 percent in the random effects model, suggesting that outliers are not a serious problem.

4 Publication Bias

If fixed and random effects estimators are both consistent for the population effect θ , why is it that they produce such different estimates of the effect size? A possible explanation is that they are differently contaminated by publication bias, to which we will turn now.

4.1 The Nature of Publication Bias

Publication bias means that journals are more likely to publish studies reporting statistically significant results. Papers reporting insignificant results are either not submitted for publication (i.e., self-censoring by the author(s)) or are rejected by the editors/referees (i.e., censoring by peers). Even though papers are not published in academic journals they may still be available as Working Papers and unpublished reports. Some authors (cf. Begg, 1994) suggest to include as many unpublished studies as possible to minimize the perverse effects



Figure 1: Funnel Plots

of publication bias. But even though it may reduce publication bias somewhat, it cannot be completely eliminated. Indeed, self-censoring by authors may be quite pernicious, which prevents them from making their findings available altogether.

Is there publication bias in our sample? To get an informal answer, we employ a funnel plot depicting the inverse of the standard error on the vertical axis and the estimated effect size on the horizontal axis. In the absence of publication bias, estimates should lie symmetrically around the true effect. The plot should look like an inverted funnel, which is wider at the bottom than at the top. Intuitively, estimates based on small samples are usually less precise and are therefore located further away from the true effect. The top panel of Figure 1 shows that estimates tend to concentrate on the right-hand side of the funnel, suggesting unidirectional publication bias (also known as type I selection bias; see Stanley, 2005). We also notice that the base of the funnel is rather wide, potentially indicating bidirectional publication bias (so-called type II selection bias). The bottom panel of Figure 1 presents an alternative funnel plot, which now measures the standard error on the vertical axis and adds 95 percent confidence bounds. It can be seen that roughly half of the data points appear to lie outside the 95 percent bands. Positive (negative) estimates seem to increase (decrease) with the standard errors, indicating that publication bias may be bidirectional.

4.2 Publication Bias Tests

We can formally test for the magnitude of publication bias while at the same time providing evidence of a genuine output effect of public capital. The starting point is equation (6), which is modified to include the standard error of each estimate $(se(\hat{\theta}_i))$ as a regressor with a view to estimating publication bias. We set up our model in a general way so as to be able to capture both fixed and random effects specifications. To differentiate the models, we introduce $\lambda_i \equiv \theta_i - \theta_0$. The fixed effects model (cf. Card and Krueger, 1995) assumes $\lambda_i = 0$ so that $\theta_i = \theta_0$, whereas the random effects model specifies $\lambda_i \sim iid(0, \sigma_{\theta}^2)$. We now get:

$$\hat{\theta}_i = \theta_0 + \lambda_i + \delta \mathrm{se}(\hat{\theta}_i) + \mu_i, \tag{9}$$

where θ_0 is the true output effect and μ_i is the error term. As discussed above, if publication selection is plaguing the meta-sample, then we should observe some relationship between the estimates and their standard errors, that is, $\delta \neq 0$. In the absence of publication bias, estimates will lie symmetrically around the true value θ_0 , which implies $\delta = 0$.

The error term μ_i in (9) is heteroscedastic given its obvious dependence on $\operatorname{se}(\hat{\theta}_i)$. In this context, OLS estimation yields inefficient (but consistent) estimates of θ and δ . In the fixed effects model, $\operatorname{se}(\hat{\theta}_i)$ provides a natural estimate of μ_i . Hence, we can divide both sides of (9) by $\operatorname{se}(\hat{\theta}_i)$ to arrive at a study's standardized effect (i.e., the *t*-value) on the left-hand side of the equation. In other words, we multiply equation (9) by $w_i \equiv 1/\operatorname{se}(\hat{\theta}_i)$.²² In the random effects model, we apply $w_i \equiv 1/\sqrt{V(e_i) + \sigma_{\theta}^2}$ to both sides of (9). Estimating the weighted version of (9) by OLS produces WLS estimates of θ_0 and δ , which are consistent and efficient.

Stanley (2005) assumes symmetric publication bias and therefore takes absolute values of the left-hand side of the weighted fixed effects version of (9) to accommodate bidirectional publication bias, that is, $\hat{\theta}_i$ is expected to be positively (negatively) correlated with $\operatorname{se}(\hat{\theta})_i$ when $\hat{\theta}_i$ is positive (negative). The more general version of Stanley's specification captures both fixed and random effects:

$$w_i|\hat{\theta}_i| = w_i \left[\theta_0 + \lambda_i + \delta \operatorname{se}(\hat{\theta}_i) + \mu_i \right].$$
(10)

Equation (10) forces the bidirectional publication bias to be symmetric, in which case δ does not depend on the sign of $\hat{\theta}_i$. It would be useful to test this restriction. Let us specify a more flexible model, in which asymmetric bidirectional publication bias is permitted:

$$w_i\hat{\theta}_i = w_i \left[\theta_0 + \lambda_i + \delta_p D_{pi} \operatorname{se}(\hat{\theta}_i) + \delta_n D_{ni} \operatorname{se}(\hat{\theta}_i) + \mu_i \right],$$
(11)

where D_{pi} (D_{ni}) is a dummy variable that equals one if $\hat{\theta}_i > 0$ $(\hat{\theta}_i < 0)$ and zero otherwise. The uniqueness of the true effect can also be tested by interacting θ_0 with the D_{pi} (D_{ni}) dummies:

$$w_i\hat{\theta}_i = w_i \left[D_{pi}(\theta_p + \lambda_{pi} + \delta_p \operatorname{se}(\hat{\theta}_i)) + D_{ni}(\theta_n + \lambda_{ni} + \delta_n \operatorname{se}(\hat{\theta}_i)) + \mu_i \right],$$
(12)

where true effect uniqueness requires imposing the condition $\theta_p = \theta_n = \theta_0$.

²²Note that in Section 3.2 the inverse variance approach was used in weighting the measurements instead of using $w_i \equiv 1/\operatorname{se}(\hat{\theta}_i)$. In terms of estimators, both weighting schemes are identical. Suppose we premultiply (9) by a $N \times N$ weighting matrix T, which features the inverse of a study's standard errors on the diagonal. Applying OLS to this equation gives the vector of estimates (including θ_0): $\delta = [\operatorname{se}(\hat{\theta})'\Omega^{-1}\operatorname{se}(\hat{\theta})]^{-1}\operatorname{se}(\hat{\theta})\Omega^{-1}\hat{\theta}$, where $\Omega^{-1} \equiv T'T$ and Ω is the variance-covariance matrix. The matrix Ω^{-1} has the inverse of the variances on the diagonal.

Equation	(9)	(10)	(12)	(11)	(11')
θ_0	0.068	0.066	_	0.064	0.064
	$(0.020)^{***}$	$(0.020)^{***}$		$(0.019)^{***}$	$(0.019)^{***}$
	[0.018]**	[0.018]***		[0.019]***	$[0.016]^{***}$
$ heta_p$	_	_	0.064	_	_
			$(0.020)^{***}$		
			[0.019]***		
$ heta_n$	—	_	0.009	_	—
			(0.031)		
			[0.010]		
δ	1.915	2.246	_	_	2.536
	$(0.522)^{***}$	$(0.482)^{***}$			$(0.371)^{***}$
	$[0.519]^{***}$	$[0.487]^{***}$			$[0.407]^{***}$
δ_p	_	_	2.565	2.576	_
•			$(0.507)^{***}$	$(0.499)^{***}$	
			$[0.529]^{***}$	$[0.524]^{***}$	
δ_n	—	_	-1.381	-2.353	_
			$(0.634)^{***}$	$(0.527)^{**}$	
			$[0.261]^{***}$	$[0.588]^{***}$	
R^2	0.356	0.371	0.707	0.706	0.706
N	76	76	76	76	76
Q-statistic	_	864.71	817.20	819.33	819.64
$\hat{\sigma}_{\theta}^2$	—	0.005	0.005	0.005	0.005
F-tests:					
$\delta_p = -\delta_n$ White	—	—	—	0.06	—
				(0.80)	
$\delta_p = -\delta_n$ Cluster	—	—	—	0.06	—
				(0.81)	
$\theta_p = \theta_n$ White	—	—	2.24	—	—
			(0.14)		
$\theta_p = \theta_n$ Cluster	—	—	5.31	—	—
			$(0.03)^{**}$		
DoF-tests:					
γ_1	0.106	0.020	-0.037	0.011	0.007
	(0.106)	(0.104)	(0.113)	(0.113)	(0.110)
2	[0.115]	[0.130]	[0.121]	[0.122]	[0.114]
R^2	0.0126	0.0004	0.0014	0.0001	0.0000

Table 2: Fixed Effects: True Effect Tests and Linear Publication Bias Correction

Notes: The values in parentheses are heteroscedasticity-robust (or White) standard errors in the case of estimates and *p*-values in the case of *F* tests. The values in brackets are clustered standard errors. ***, **, * denote significance at the 1, 5, and 10 percent level, respectively. The dependent variable in the degrees of freedom (DoF) test is the logarithm of the *t*-statistic corrected for publication bias. To save on space, the constant (γ_0) of the DoF test is not reported.

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$						
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Equation	(9)	(10)	(12)	(11)	(11')
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	θ_0	0.085	0.089	_	0.087	0.081
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		$(0.022)^{***}$	$(0.020)^{***}$		$(0.020)^{***}$	$(0.014)^{***}$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		$[0.024]^{***}$	$[0.025]^{***}$		$[0.025]^{***}$	$[0.014]^{***}$
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ heta_p$	—	—	0.092	—	—
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	•			$(0.021)^{***}$		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$				[0.025]***		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ heta_n$	_	_	-0.035	_	_
$\begin{split} \delta & 1.443 & 1.666 & - & - & 2.066 \\ & (0.391)^{***} & (0.349)^{***} & (0.233)^{*:} \\ & [0.379]^{***} & [0.325]^{***} & [0.214]^{*:} \\ \delta_p & - & - & 1.872 & 1.943 & - \\ & & (0.356)^{***} & (0.347)^{***} \\ & [0.352]^{***} & [0.348]^{***} \\ \delta_n & - & - & -0.680 & -2.399 & - \\ & (0.793) & (0.534)^{***} \\ \hline R^2 & 0.589 & 0.715 & 0.748 & 0.742 & 0.741 \\ N & 76 & 76 & 76 & 76 & 76 \\ F-tests: \\ \delta_p &= -\delta_n White & - & - & 0.37 & - \\ & & & & & & & & & & \\ \delta_p &= -\delta_n Cluster & - & - & 0.28 & - \\ & & & & & & & & & & & & \\ \theta_p &= \theta_n White & - & - & 5.01 & - & - \\ & & & & & & & & & & & & \\ \theta_p &= \theta_n White & - & - & 5.01 & - & - \\ & & & & & & & & & & & & \\ \theta_p &= \theta_n Cluster & - & - & 6.68 & - & - \\ & & & & & & & & & & & \\ 0.02)^{**} & DoF-tests: \\ \gamma_1 & & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 \\ \end{array}$				(0.053)		
$ \begin{split} \delta & 1.443 & 1.666 & - & - & 2.066 \\ & (0.391)^{***} & (0.349)^{***} & (0.233)^{**} \\ & [0.379]^{***} & [0.325]^{***} & [0.214]^{**} \\ & [0.352]^{***} & (0.347)^{***} \\ & [0.352]^{***} & [0.348]^{***} \\ & & [0.352]^{***} & [0.348]^{***} \\ & & & [0.352]^{***} & [0.348]^{***} \\ & & & & & & & \\ & & & & & & \\ &$				[0.030]		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	δ	1.443	1.666	_	_	2.066
$\begin{split} & \begin{bmatrix} [0.379]^{***} & [0.325]^{***} & & \\ [0.379]^{***} & \begin{bmatrix} [0.325]^{***} & & \\ [0.352]^{***} & \\ (0.347)^{***} & \\ & \\ & & \\ & \\ & \\ & \\ & \\ & \\ & \\ $		$(0.391)^{***}$	$(0.349)^{***}$			$(0.233)^{***}$
$\begin{split} \delta_p & - & - & - & 1.872 & 1.943 & - & \\ & & & & (0.356)^{***} & (0.347)^{***} & \\ & & & & & [0.352]^{***} & [0.348]^{***} & \\ \delta_n & - & - & - & -0.680 & -2.399 & - & \\ & & & & & (0.793) & (0.534)^{***} & \\ \hline R^2 & & 0.589 & 0.715 & 0.748 & 0.742 & 0.741 & \\ \hline N & & 76 & 76 & 76 & 76 & 76 & \\ \hline F\text{-tests:} & & & & \\ \delta_p &= -\delta_n \text{ White} & - & - & - & 0.37 & - & \\ & & & & & (0.54) & \\ \delta_p &= -\delta_n \text{ Cluster} & - & - & - & 0.28 & - & \\ & & & & & & (0.60) & \\ \theta_p &= & \theta_n \text{ White} & - & - & - & 0.28 & - & \\ & & & & & & (0.60) & \\ \theta_p &= & \theta_n \text{ White} & - & - & - & 0.28 & - & \\ & & & & & & & (0.60) & \\ \theta_p &= & \theta_n \text{ White} & - & - & - & 0.028 & - & \\ & & & & & & & (0.60) & \\ \theta_p &= & \theta_n \text{ Cluster} & - & - & 6.68 & - & - & \\ & & & & & & (0.02)^{**} & \\ \hline \text{DoF-tests:} & & & & \\ \gamma_1 & & & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 & \\ & & & & & & (0.110) & & (0.110) & \\ \end{array}$		$[0.379]^{***}$	$[0.325]^{***}$			[0.214]***
$\begin{split} \delta_n & - & - & - & \begin{pmatrix} (0.356)^{***} & (0.347)^{***} \\ [0.352]^{***} & [0.348]^{***} \\ [0.352]^{***} & [0.348]^{***} \\ [0.394]^{**} & [0.534)^{***} \\ [0.394]^{**} & [0.618]^{***} \\ \hline R^2 & 0.589 & 0.715 & 0.748 & 0.742 & 0.741 \\ \hline N & 76 & 76 & 76 & 76 & 76 \\ \hline F\text{-tests:} & & & & & & & \\ \delta_p &= -\delta_n \text{ White} & - & - & - & 0.37 & - & & & & \\ \delta_p &= -\delta_n \text{ Cluster} & - & - & - & 0.28 & - & & & & & \\ \delta_p &= -\delta_n \text{ Cluster} & - & - & - & 0.28 & - & & & & & \\ \theta_p &= \theta_n \text{ White} & - & - & - & 6.68 & - & - & & & & & \\ \theta_p &= \theta_n \text{ Cluster} & - & - & 6.68 & - & - & & & & & & \\ 0.02)^{**} & & & & & & & & \\ DoF\text{-tests:} & & & & & & & & & \\ \gamma_1 & & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 & & & & & & & \\ 0.020 & & & & & & & & & & & & & & \\ 0.020 & & & & & & & & & & & & & & \\ \end{array}$	δ_{p}	_	_	1.872	1.943	_
$\begin{split} \delta_n & - & - & - & \begin{bmatrix} 0.352 \end{bmatrix}^{***} & \begin{bmatrix} 0.348 \end{bmatrix}^{***} \\ -0.680 & -2.399 & - \\ & (0.793) & (0.534)^{***} \\ \hline \\ R^2 & 0.589 & 0.715 & 0.748 & 0.742 & 0.741 \\ \hline N & 76 & 76 & 76 & 76 & 76 \\ \hline \\ F\text{-tests:} & & & & & \\ \delta_p &= -\delta_n \text{ White} & - & - & - & 0.37 & - \\ & & & & & (0.54) \\ \delta_p &= -\delta_n \text{ Cluster} & - & - & - & 0.28 & - \\ & & & & & & (0.60) \\ \theta_p &= & \theta_n \text{ White} & - & - & - & 5.01 & - & - \\ & & & & & & (0.60) \\ \theta_p &= & \theta_n \text{ White} & - & - & - & 6.68 & - & - \\ & & & & & & (0.02)^{**} \\ \hline \\ DoF\text{-tests:} & & & & \\ \gamma_1 & & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 \\ & & & & & (0.112) \\ \hline \end{split}$	P			$(0.356)^{***}$	$(0.347)^{***}$	
$\begin{split} \delta_n & - & - & - & -0.680 & -2.399 & - \\ & & & & & & & & \\ & & & & & & & \\ & & & & & & & \\ \hline 0.793) & & & & & & \\ & & & & & & & \\ \hline 0.534)^{***} & & & & \\ \hline 0.0589 & & & & & & \\ \hline N & & & & & & & \\ \hline R^2 & & & & & & \\ \hline R^2 & & & & & & \\ \hline R^2 & & & & & & \\ \hline N & & & & & & \\ \hline N & & & & & & & \\ \hline N & & & & & & & \\ \hline F^- \text{tests:} & & & & \\ \delta_p & = & -\delta_n \text{ White} & - & & & & & \\ \delta_p & = & -\delta_n \text{ White} & - & & & & & \\ \hline \delta_p & = & -\delta_n \text{ Cluster} & - & & & & & \\ & & & & & & & \\ \delta_p & = & -\delta_n \text{ White} & - & & & & & \\ \hline \delta_p & = & -\delta_n \text{ White} & - & & & & & \\ \delta_p & = & -\delta_n \text{ White} & - & & & & & \\ \delta_p & = & -\delta_n \text{ White} & - & & & & & \\ \delta_p & = & -\delta_n \text{ Cluster} & - & & & & \\ \hline \delta_p & = & \theta_n \text{ White} & - & & & & & \\ \delta_p & = & \theta_n \text{ Cluster} & - & & & & \\ \hline \delta_p & = & \theta_n \text{ Cluster} & - & & & & & \\ \delta_p & = & \theta_n \text{ Cluster} & - & & & & & \\ \hline 0.05^+ \text{tests:} & & & \\ \gamma_1 & & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 \\ \hline (0.102) & & & & & \\ \hline 0.110) & & & & & \\ \hline 0.110) & & & & \\ \hline \end{array}$				[0.352]***	[0.348]***	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	δ_n	_	_	-0.680	-2.399	_
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	10			(0.793)	$(0.534)^{***}$	
R^2 0.589 0.715 0.748 0.742 0.741 N 76 76 76 76 76 76 76 F-tests: $\delta_p = -\delta_n$ White - - 0.37 - (0.54) - $\delta_p = -\delta_n$ Cluster - - 0.37 - (0.54) - - - (0.54) - - - (0.54) - - - (0.60) - - - (0.60) - - - - (0.60) -				[0.394]*	[0.618]***	
N 76<	R^2	0.589	0.715	0.748	0.742	0.741
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	N	76	76	76	76	76
$\begin{split} \delta_p &= -\delta_n \text{ White } & - & - & - & 0.37 & - \\ & & & & & & & \\ \delta_p &= -\delta_n \text{ Cluster } & - & - & - & 0.28 & - \\ & & & & & & & \\ \theta_p &= & \theta_n \text{ White } & - & - & 5.01 & - & - \\ \theta_p &= & \theta_n \text{ Cluster } & - & - & 6.68 & - & - \\ & & & & & & \\ \theta_p &= & \theta_n \text{ Cluster } & - & - & 6.68 & - & - \\ & & & & & & \\ 0.05^{+} \text{tests: } & & & \\ \gamma_1 & & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 \\ & & & & & & & \\ (0.02)^{**} & & & & & \\ 0.057 & & & & & & \\ 0.057 & & & & & & \\ (0.112) & & & & & & & \\ 0.112) & & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & & & \\ 0.112) & & \\ 0.112) & & & \\ 0.112) & & & \\ 0.112) & & & \\ 0.112) & & \\ 0.112) & & & \\ 0.112) & $	F-tests:					
$\begin{split} \delta_p &= -\delta_n \text{ Cluster} & - & - & - & 0.28 & - \\ \theta_p &= & \theta_n \text{ White} & - & - & 5.01 & - & - \\ \theta_p &= & \theta_n \text{ Cluster} & - & - & 6.68 & - & - \\ \theta_p &= & \theta_n \text{ Cluster} & - & - & 6.68 & - & - \\ 0.02)^{**} & & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 \\ 0.082) & & (0.102) & & (0.112) & (0.110) \\ \end{split}$	$\delta_n = -\delta_n$ White	_	_	_	0.37	_
$\delta_{p} = -\delta_{n} \text{ Cluster} \qquad - \qquad - \qquad - \qquad 0.28 \qquad - \qquad (0.60)$ $\theta_{p} = \theta_{n} \text{ White} \qquad - \qquad - \qquad 5.01 \qquad - \qquad - \qquad (0.03)^{**}$ $\theta_{p} = \theta_{n} \text{ Cluster} \qquad - \qquad - \qquad 6.68 \qquad - \qquad - \qquad (0.02)^{**}$ DoF-tests: $\gamma_{1} \qquad 0.066 \qquad 0.057 \qquad -0.022 \qquad 0.088 \qquad 0.103 \qquad (0.110) \qquad (0.110)$	P				(0.54)	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\delta_n = -\delta_n$ Cluster	_	_	_	0.28	_
$ \theta_p = \theta_n \text{ White } 5.01 (0.03)^{**} \\ \theta_p = \theta_n \text{ Cluster } 6.68 (0.02)^{**} \\ \hline \text{DoF-tests:} \\ \gamma_1 = 0.066 & 0.057 & -0.022 & 0.088 & 0.103 \\ (0.082) = (0.102) & (0.112) & (0.110) \\ \hline \end{array} $	P				(0.60)	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\theta_n = \theta_n$ White	_	_	5.01	_	_
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	P R			$(0.03)^{**}$		
$\begin{array}{c} & (0.02)^{**} \\ \hline \text{DoF-tests:} \\ \gamma_1 & 0.066 & 0.057 & -0.022 & 0.088 & 0.103 \\ (0.082) & (0.102) & (0.112) & (0.110) \\ \end{array}$	$\theta_n = \theta_n$ Cluster	_	_	6.68	_	_
DoF-tests: γ_1 0.066 0.057 -0.022 0.088 0.103 $(0, 082)$ $(0, 102)$ $(0, 112)$ $(0, 110)$ $(0, 110)$	p n			$(0.02)^{**}$		
γ_1 0.066 0.057 -0.022 0.088 0.103 (0.082) (0.102) (0.112) (0.110) (0.110)	DoF-tests:			× /		
	γ_1	0.066	0.057	-0.022	0.088	0.103
(0.082) (0.103) (0.112) (0.110) (0.119)	1 -	(0.082)	(0.103)	(0.112)	(0.110)	(0.119)
[0.087] $[0.100]$ $[0.116]$ $[0.117]$ $[0.124]$		[0.087]	[0.100]	[0.116]	[0.117]	[0.124]
R^2 0.0082 0.0036 0.0005 0.0082 0.0093	R^2	0.0082	0.0036	0.0005	0.0082	0.0093

Table 3: Random Effects: True Effect Tests and Linear Publication Bias Correction

Notes: The values in parentheses are heteroscedasticity-robust (or White) standard errors in the case of estimates and *p*-values in the case of *F* tests. The values in brackets are clustered standard errors. The *Q*-statistic and $\hat{\sigma}^2$ can be found in Table 2. ***, **, * denote significance at the 1, 5, and 10 percent level, respectively. The dependent variable in the degrees of freedom (DoF) test is the logarithm of the *t*-statistic corrected for publication bias. To save on space, the constant (γ_0) of the DoF test is not reported.

An alternative way to test whether a genuine empirical effect exists is to relate a study's *t*-value to its degrees of freedom (df):

$$\ln \left| \frac{\hat{\theta}_i^*}{\operatorname{se}(\hat{\theta}_i)} \right| = \gamma_0 + \gamma_1 \ln \mathrm{df}_i + \mu_i, \tag{13}$$

where $\hat{\theta}_i^*$ is $\hat{\theta}_i$ corrected for publication bias and γ_0 and γ_1 are parameters to be estimated. If we cannot reject the hypothesis that $\gamma_1 = 0$, then either the true effect is zero or publication bias is too severe. A genuine empirical effect implies $\gamma_1 = 1/2$ (cf. Card and Krueger, 1995).

4.3 Estimation Results

Tables 2–3 present the estimates of the true effect and the size of publication bias of fixed effects and random effects models, respectively. Besides White standard errors, which correct for identified heteroscedasticity, we present clustered standard errors. For this purpose, we have defined 27 clusters (see Appendix A.1 for a description). The results show, however, that clustering does not affect our results much.²³ Column (1) estimates equation (9), yielding $\theta = 0.068$ in the fixed effects model and $\theta = 0.085$ in the random effects model. Both are highly significant. As indicated by the significant value of δ —and suspected from Figure 1 publication bias proves to be very significant in both models. Intuitively, even studies with very small sample sizes—and thus high standard errors of the estimates—manage to find a high enough θ such that a significant result is obtained. The fixed effects specification yields a larger parameter for publication bias than the random effects model. Not surprisingly, the estimate of δ with symmetric publication bias imposed (see column (2)) turns out to be significant in both models. Imposing symmetry raises the publication bias coefficient slightly.

Column (3) shows that the estimated true effect for positive estimates, θ_p , is positive, highly significant, and very much in line with previous estimates of θ . Perhaps more surprisingly, the estimate of θ_n is also positive in the fixed effects model, though not statistically different from zero, suggesting that negative estimates in the meta-sample are very likely to be just the result of a mixture of pure chance and publication selection. In the random effects model, θ_n is negative, but again insignificant. Column (4) imposes a symmetric true effect

 $^{^{23}}$ Clustering of measurements makes more of a difference in the meta-regression analysis of Section 5.

and shows estimates of positive and negative publication bias parameters, δ_p and δ_n , which appears quite symmetric; the *F*-test of the hypothesis $\delta_p = -\delta_n$ is not rejected. The hypothesis that publication bias is unidirectional (i.e., $\delta_p = \delta_n$) is naturally rejected (the *F*-test statistic is not reported). Imposing symmetric publication bias and true effect uniqueness (which we label (11'), see column (5)) does not change the point estimate of θ much, but increases the magnitude of publication bias somewhat.

Estimating (13) without publication bias correction yields an insignificant value of the γ_1 coefficient in both models (results are not reported), for which three potential explanations exist. First, a genuine empirical effect may be absent. The second reason is the presence of strong heterogeneity. Finally, γ_1 can be insignificant because of substantial publication bias, evidence of which has been presented above. To account for publication bias, we re-estimate equation (13) with the left-hand side variable corrected for publication bias (see the bottom panel of Tables 2–3). All γ'_1 's remain insignificant, however. Section 5 demonstrates that this is caused by omitted variables.

We can now use the estimate of δ from the last column of Tables 2–3 to correct the meta- θ for linear publication bias (see the right panel of Table 1). As compared to the case without publication bias correction, the difference between fixed and random effects estimates narrows, where the random effects estimates show a larger decline than the fixed effects estimates. As before, the *Q*-test strongly favors the random effects estimator. For the entire sample, we find a random effects meta- θ of 0.081, lying within the 95 percent confidence interval. Focusing on studies using the pure production function only, does not affect the estimate of θ much, whereas exclusion of VAR-based studies decreases it to 0.075.

5 Meta-Regression Analysis

The previous section showed that the true output elasticity of public capital is heterogeneous. However, heterogeneity has not been explicitly modeled yet. This section employs a metaregression approach to find the determinants of the excess variation of estimates of the output elasticity of public capital across studies.

5.1 Methodological Framework

The starting point of the meta-regression model is the assumption that the true effect being estimated depends on K study-specific factors. Table 4 provides the complete list of moderators that are taken into consideration. In the context of symmetric bidirectional publication bias, we employ an extended version of (12) capturing both fixed and random effects:

$$w_i\hat{\theta}_i = w_i \left(\theta_0 + \lambda_i + \sum_{j=1}^M \phi_j D_{ji} + \delta \operatorname{se}(\hat{\theta}_i) (D_{pi} - D_{ni}) + \mu_i \right),$$
(14)

where ϕ'_{j} s are estimated coefficients and D_{ji} (for j = 1, 2, ..., K) is a dummy variable that equals one if the *i*-th estimate is obtained from a study described by characteristic j and zero otherwise. We include country dummies to control for dependency across measurements for the same country. The dummies ensure that the within (and not the between) variation is captured across estimates.²⁴

We estimate equation (14) by OLS, using the weights defined earlier, which gives rise to WLS estimates of the parameters. Because the latter do not take into account the covariances across error terms they represent a partial GLS estimator. We apply a general-to-specific approach to reduce the model to a parsimonious specification for two reasons: (i) many of the moderator variables are not significantly different from zero; and (ii) the small number of degrees of freedom (in our case 42) if all moderator variables are included. In each step of the procedure, we delete the least significant moderator variable until we are left with a model in which the variables are significant at least at the 10 percent level. Subsequently, we test for: (i) the joint insignificance of the excluded variables; and (ii) publication bias symmetry.

Because of the relatively large number of studies on the United States (41 percent of total), a few studies use identical data sets. For example, Munnell (1990b), Eisner (1991), and Baltagi and Pinnoi (1995) all use data for the 1970–1986 period. Other studies on the United States use slightly different sample periods, thereby yielding very similar data sets. Overlapping data sets generate sample dependency. For some smaller European countries, notably Italy and Spain, we also observe some overlap in data sets, again creating dependency

²⁴We include country dummies D_{ji} (for j = K + 1, ..., M) if there at least two measurements for a single country. This yields M - K - 1 = 9 country dummies.

Empirical Dimension	Dummy Variable	Definition (1 if, 0 otherwise)
Definition of output	Private	Dependent variable is private sector output at country or state level
Country fixed effect	Country i	Data are for country i^a
	Developing	Data are for developing countries
Type of public capital	Core	Core capital is used ^b
	Transport	Transportation capital is considered
	Proxy	Capital is proxied by the public investment-to-GDP ratio
	Reg-loc	Only regional/local capital is considered
Data aggregation	Reg-da	Regional data is used
Empirical model	Pfrontier	A production frontier approach is employed
	$\operatorname{PF-based}$	A transformation of a production function is used
	Gregr	A growth regression is considered
Econometric specification	Trend	A time trend is included
	Cap-util	Capacity utilization is controlled for
	CRTS	Constant returns to scale is imposed
Estimation method	Spurious	Equation is estimated in levels without cointegration test
	Coint-all	Cointegration relationship is found using one of the available cointegration methods
	Coint-VAR	Cointegration relationship is studied using Johansen's VAR analysis
	FE-levels	Unit-specific fixed effects with variables in levels are employed
	TE-levels	Time effects with variables in levels are employed
	FE-dif	Unit-specific fixed effects with variables in differences are employed
	TE-dif	Time effects with variables in differences are employed
	IV-GMM	IV or GMM estimators are used
	AR	An autoregressive residual correction is applied
aCountry-specific fixed effects are	e included for those count	ries for which there is more than one estimate: Australia, Canada, France, Germany, Italy, Japan,
Mexico, Spain, and United States	s; and b Core capital is i	t subset of non-military public capital consisting of two main components: transportation capital

(highways, mass transit, etcetera) and public utilities (water and sewers, electrical and gas facilities, etcetera).

Table 4: List of Moderators Used in the Meta-Regression Model

Equation	Partial GLS I				Full GLS
	Fixed Eff	fects	Random E	ffects	RE
	White	Cluster	White	Cluster	White
θ_0	0.120***	0.120***	0.125***	0.127***	0.086***
Ŭ	(0.008)	[0.009]	(0.017)	[0.022]	(0.036)
Developing	-0.068***	-0.068***	-0.070***	-0.118***	-0.117***
	(0.023)	[0.023]	(0.025)	[0.024]	(0.043)
Core	0.073**	0.073	0.085^{*}	0.064	0.078*
	(0.036)	[0.049]	(0.050)	[0.069]	(0.047)
Reg-loc	0.057**	0.057	0.066**	0.079**	0.075*
	(0.025)	[0.023]	(0.030)	[0.031]	(0.044)
Transport	_	-	_	-0.051*	_
				[0.028]	
Prox	—	—	—	0.082^{**}	0.103^{*}
				[0.031]	(0.055)
Private	—	-	—	_	-0.046
					(0.027)
Reg-da	-0.085***	-0.085***	-0.081***	-	-0.119***
	(0.018)	[0.014]	(0.027)		(0.040)
Pfrontier	0.099^{***}	0.099***	0.088^{**}	0.075^{***}	0.132^{***}
	(0.029)	[0.033]	(0.037)	[0.024]	(0.049)
Gregr	-0.085***	-0.085***	-0.092***	-0.171^{***}	-0.148**
	(0.015)	[0.010]	(0.023)	[0.032]	(0.057)
PF-based	-	-	-	-0.059**	-
				[0.024]	
Trend	-0.039*	-0.039	-0.049*	-0.045**	-0.050*
	(0.020)	[0.026]	(0.026)	[0.020]	(0.030)
Spurious	0.051*	0.051^{***}	0.060*	0.090***	0.118**
	(0.028)	[0.016]	(0.033)	[0.016]	(0.045)
Coint-all	0.124^{***}	0.124^{***}	0.105^{**}	0.158^{***}	0.096^{***}
	(0.034)	[0.043]	(0.047)	[0.036]	(0.024)
Coint-VAR	0.066*	0.066	0.072	-	0.084^{**}
	(0.039)	[0.047]	(0.056)		(0.038)
F'E-levels	0.040*	0.040	0.039	-	0.059*
	(0.021)	[0.034]	(0.027)		(0.031)
IV-GMM	-	-	-	0.028**	-
apma				[0.012]	0.05044
CRTS	—	—	—	—	0.058**
5			4 04 04 44	1 000****	(0.025)
δ	1.779***	1.779***	1.818***	1.823***	1.514***
	(0.356)	[0.390]	(0.345)	[0.375]	(0.555)
R ²	0.933	0.933	0.872	0.874	0.776
N	76	76	76	76	76
Q_{a^2}	186.455	186.455	-	-	-
σ_{θ}^2	0.001	0.001	-	_	_
r-tests:	1.010	1 1 1 0	1 000	1 470	0.000
$o_p = -o_n$	1.910	1.110	1.280	1.4(0)	0.000
4 4 0	(0.173)	(0.302)	(0.263)	(0.237)	(0.999)
$\varphi_1 = \dots = \phi_X = 0$	1.340	1.300	1.300	1.390	0.420
D.E.t.sta	(0.235)	(0.251)	(0.141)	(0.101)	(0.929)
DOF tests:	0.950***	0.950***	0.000***	0.000***	0 201***
γ_1	0.350^{-+++}	0.350****	0.280^{-+++}	0.280****	$0.301^{}$
	(0.102)	[0.076]	(0.075)	[0.050]	(0.103)

 Table 5: Meta-Regression Results

Notes: The values in parentheses are heteroscedasticity-robust (or White) standard errors in the case of estimates and *p*-values in the case of *F*-tests. ***, **, * denote significance at the 1, 5, and 10 percent level, respectively. The random effects (RE) models (both partial and full GLS) use the $\hat{\sigma}_{\theta}^2$ obtained from the fixed effects partial GLS model. The second *F*-test in the table tests whether or not *X* excluded variables are jointly insignificant. The dependent variable in the degrees of freedom (DoF) test is the logarithm of the *t*-statistic corrected for publication bias. See equation (13). To save on space, the constant (γ_0) of the DoF test is not reported. across studies. To address the sample overlap problem, we make use of full GLS estimation (Appendix A.2), which is applied to the random effects model. The GLS estimator weights the original measurements (our dependent variable) and the explanatory variables by a fully specified variance-covariance matrix. The degree of sample overlap is used to recover the off-diagonal elements of the variance-covariance matrix, which were assumed to be zero in the analysis of Sections 4.2–4.3. Because the information that is used to construct the clusters is basically the same as that needed to estimate the off-diagonal elements of the variance-covariance matrix, we do not use clusters in applying the full GLS estimator.

5.2 Regression Results

The last column of Table 5 presents the regression results for the random effects model estimated by full GLS, which we consider the benchmark model. The estimate of the true output elasticity of public capital is 0.086, which is slightly larger than the value of 0.081 found in the true effect tests of Table 3. Note that the parameter θ_0 is now measuring the true effect conditional on a given set of study characteristics, whereas in (9) it measures the unconditional true effect. Publication bias is significant and has the expected sign. It is, however, much smaller than the value found in the true effect tests (compare column (5) of Tables 3 and 5). The symmetry of publication bias could not be rejected. In general, the publication bias parameter becomes more precisely estimated and rises slightly in size when we impose the symmetry restriction.

We find 13 moderator variables that are significant (at least at the 10 percent level). Six of these variables are related to the estimation method and econometric specification. Studies allowing for a time trend find smaller output elasticities of public capital. A larger estimate of θ is found for studies: (i) estimating a model in levels that is likely to be spurious; (ii) including fixed effects in a levels specification; (iii) finding cointegration based on various methods, such as Engle and Granger's (1987) or Johansen's (1988) method; and (iv) imposing a CRTS restriction across all inputs. The coefficient of the *Coint-all* variable (measuring a cointegrating relationship among variables using various methods) is very large; it even exceeds the value of the conditional true effect. In addition, the coefficient of potentially spurious regressions (i.e., *Spurious*) is also large, partially explaining why the early literature found unreasonably large estimates. Many of the latter papers may also contain unidentified cointegrating relationships among the variables.

In line with the empirical partial results of Section 4.3, we find a significantly positive coefficient for studies employing core public capital, implying that railways and airports are more productive than public office buildings. In contrast, we find a significantly negative coefficient for studies using regional data, supporting the presence of spillover effects at the regional level that cannot be internalized at the national level. Public capital that is provided by local/regional governments seems to be more productive than that at the national level. Local governments apparently can better target public spending than national governments. The coefficient of the *Proxy* variable is significantly positive at the 10 percent level. Studies proxing public capital by the public investment-to-GDP ratio find larger output elasticities of public capital.

The type of empirical model affects the results too. Studies employing growth regressions yield smaller output elasticities than those employing a pure production function approach. Intuitively, the former is quite flexible in terms of the set of explanatory variables, which may include those variables that are strongly correlated with public capital.²⁵ Consequently, some of the variance of our dependent variable may be picked up by variables other than public capital. Studies based on the production frontier approach yield larger coefficients, however.

The specification incorporates country dummies to control for country-specific effects. Five of the nine country dummies (which are not reported) are significantly different from zero. Surprisingly, the dummy for the United States—found to be significant in Button's (1998) study—is only significant at the 10 percent level. Countries that are larger in size (in terms of surface area) seem to have smaller coefficients of the country dummies.²⁶ The dummy variable for developing countries is significantly negative, suggesting that the output elasticity of public capital for this country group is lower than for industrialized countries. In view of the typically low public capital-output ratios of developing countries, it is very likely

 $^{^{25}}$ Durlauf et al. (2005) argue that there is an abundance of potential growth determinants. Roughly 145 different proxies for growth determinants have been used in cross-country growth regressions.

 $^{^{26}}$ Regressing the estimated coefficients of the country dummies on the natural logarithm of country size yields an estimated slope coefficient of -0.168 and a R^2 of 0.44.

that their public capital is contributing more to productivity than in industrialized countries.

The R^2 of 0.78 shows that the model fits the data fairly well. Compared with the univariate true effect tests of Section 4, the inclusion of moderator variables explains some additional variation. Previously, some of this variation was picked up by publication bias, which plays a larger role in the univariate true effect tests. The *F*-test of joint insignificance of the excluded moderators cannot be rejected. Note that the degrees of freedom test now shows a significant γ_1 coefficient; it is not statistically different from 0.5 at the 5 percent level, providing evidence of a true effect.

5.3 Robustness Analysis

Columns (1)–(4) of Table 5 show the results obtained by applying the partial GLS estimator to the fixed effects and random effects models. Both regressions yield a larger true effect and a larger publication bias coefficient than in the full GLS specification. In addition, R^2 is larger than in the full GLS specification. Apparently, more variation is attributed to publication bias. Like before, we cannot reject the symmetry of publication bias. Dropping the countryspecific fixed effects from the equation with White standard errors does not affect the size of the true effect. In the case of clustered errors, the true effect rises by 0.02 (fixed effects) and 0.01 (random effects) if the country-specific fixed effect is left out.

As compared with the full GLS specification, the set of significant moderator variables changes somewhat in the random effects specification with clustered errors. Five moderator variables are no longer significant: core public capital, regional data, Johansen's cointegration method, fixed effects estimation in levels, and the CRTS restriction. Three new significant variables enter (i.e., transportation capital, production function based studies, and IV estimation), potentially reflecting the effect of collinearity among the moderator variables.²⁷ In the fixed effects model with clustered standard errors, which is not our preferred specification, even more variables drop out. Quite some overlap exists with the dropped variables in the random effects model.

²⁷Correlations above 0.75 were recorded between *Private* and *CRTS*, *IV-GMM* and *Private*, *Private* and *Spurious*, *Reg-da* and the US dummy, *Transport* and the US dummy, and *Spurious* and the Australia dummy.

6 Conclusions

This paper has assessed the output-elasticity of public capital by means of several metaanalytical techniques, including meta-analysis and meta-regression analysis. A sample was gathered of 76 studies focusing on the production function approach in a broad sense. The estimates of the output elasticity of public capital were corrected for publication bias. In the meta-regression analysis, both fixed and random effects models were estimated using Weighted Least Squares (or partial Generalized Least Squares). To address sample overlap across studies, a full Generalized Least Squares estimator was employed in the random effects model.

Publication bias is shown to be substantial and significant in both the fixed and random effects model of the simple meta-analysis. Moreover, publication bias appears to be bidirectional and symmetric. The true output elasticity of public capital is positive and significant no matter whether publication bias is corrected for. After correcting for publication bias, the output elasticity of public capital in the (preferred) random effects model turns out to be 0.081 compared with 0.064 in the fixed effects model.

Heterogeneity across estimates of the output elasticity of public capital is predominantly caused by differences in research design, such as the econometric specification, estimation technique, type of empirical model, type of public capital, and aggregation level of public capital. On the chosen estimation technique and econometric specification, the analysis indicates that larger meta-output elasticities are obtained in studies featuring cointegration, spurious relationships among variables, levels specifications estimated by fixed effects, and constant returns to scale restrictions. Smaller meta-output elasticities are found in studies employing time trends. The type of empirical model matters too. Production frontier studies yield larger output elasticities, whereas growth regressions give rise to smaller estimates of the output elasticity. On the kind of public capital and public capital data, the analysis indicates that core public capital features a larger output elasticity than specifications drawing on a more broadly defined concept of public capital. Studies employing regional data yield smaller estimates of the output elasticity of public capital, whereas studies focusing on locally installed public capital find a higher output elasticity than those dealing with centrally owned public capital.

Unlike Aschauer's estimate of 0.39, the meta-regression analysis yields a reasonable metaoutput elasticity of public capital of 0.086 (which is slightly larger than the value found in the univariate true effect test). This implies a marginal productivity of public capital of roughly 17 percent in 2001.²⁸ In that same year, the real long-term rate of interest which is assumed to reflect the marginal productivity of private capital in a perfect market amounted to 2.6 percent, suggesting that investment in public capital should be encouraged from a macroeconomic point of view. In view of the above meta-regression results, the large estimates found in the early studies seem to be caused by either unidentified (but present) cointegrating relationships or spurious relationships among national time-series data.

Our analysis provides clear evidence in favor of public capital investment, but it is not without limitations. The Cobb-Douglas production function, on which the majority of studies is based, is quite restrictive. Extending the analysis by allowing for the much more flexible translog function is certainly an improvement. A key issue here is that output elasticities of public capital have to be calculated from the translog parameters. Although many studies carry out such computations for the elasticities themselves, few do so for the standard errors. Another fruitful research avenue is to combine the estimates from studies using the behavioral approach, that is, cost and/or profit functions, with those of the production function. To the best of our knowledge, such work has not been undertaken yet. Finally, in future work we intend to correct our meta-estimate of the elasticity of public capital for nonlinear publication bias.

 $^{^{28}}$ In the United States, the public capital-to-GDP ratio amounted to 50.6 percent in 2001. See Kamps (2006).

Appendix

A.1 Constructing Clusters

In constructing the clusters, we took note of the following guiding principles. First, the number of clusters must be as large as possible given the size of the data set. Some authors, for example, Cameron et al. (2007), argue that at least 50 clusters are needed for accurate inference. In that case, the cluster robust standard error converges to the true standard error. Second, cluster sizes should be rather balanced. It would be natural to construct the clusters on the basis of countries. Given that we have only 13 single country studies, differing substantially in terms of the number of estimates, a small number of unbalanced clusters would result. For instance, the United States alone would yield a cluster with almost half of the sample, whereas many other countries would have one estimate only. In this context, correcting the standard errors by cluster would probably do more harm than good.

To arrive at a sufficient number of clusters, we define various clusters within the same country, especially for the United States. In defining the clusters, we use the degree of overlap of the samples underlying the studies. We allocate studies to a single cluster if the degree of sample overlap exceeds the pre-specified threshold of 80 percent. In a few cases (e.g., Ratner, 1983), we prefer to accommodate studies below the threshold in existing clusters rather than to isolate them in a one-study cluster. We thus employ 27 clusters, ranging from a minimum of one to a maximum of 14 estimates per cluster. On average, a cluster consists of 2.8 observations.

A.2 The Full GLS Estimator

To arrive at the full GLS estimator, we start with the following random effects model:

$$\hat{\theta}_{i} = \theta_{0} + \lambda_{i} + \sum_{j=1}^{M} \phi_{j} D_{ji} + \delta \operatorname{se}(\hat{\theta}_{i}) (D_{pi} - D_{ni}) + \mu_{i},$$
(A.1)

which can be written in compact form as:

$$\hat{\theta} = \mathbf{X}\psi + \mathbf{v},\tag{A.2}$$

where ${\bf X}$ is defined as:

$$\mathbf{X} \equiv \begin{bmatrix} 1 & D_{11} & \dots & D_{M1} & \operatorname{se}(\hat{\theta}_1)(D_{p1} - D_{n1}) \\ 1 & D_{12} & D_{M2} & \operatorname{se}(\hat{\theta}_2)(D_{p2} - D_{n2}) \\ \vdots & \vdots & \ddots & \vdots & & \vdots \\ 1 & D_{1N} & \dots & D_{MN} & \operatorname{se}(\hat{\theta}_N)(D_{pN} - D_{nN}) \end{bmatrix},$$
(A.3)

and $\hat{\theta}$, ψ , and **v** are

$$\hat{\theta} = \begin{bmatrix} \hat{\theta}_1 \\ \hat{\theta}_2 \\ \vdots \\ \hat{\theta}_N \end{bmatrix}, \quad \psi = \begin{bmatrix} \theta_0 \\ \phi_1 \\ \vdots \\ \phi_M \\ \delta \end{bmatrix}, \quad \mathbf{v} \equiv \lambda + \mathbf{u} = \begin{bmatrix} \lambda_1 + u_1 \\ \lambda_2 + u_2 \\ \vdots \\ \lambda_N + u_N \end{bmatrix}.$$
(A.4)

Note that $E(\mathbf{v}) = \mathbf{0}$ and $V(\mathbf{v})$ (denoted by $\mathbf{\Omega}$) are defined as:

$$\boldsymbol{\Omega} \equiv \begin{bmatrix} \sigma_{\theta}^2 + \sigma_{u_1}^2 & \sigma_{u_2u_1} & \cdots & \sigma_{u_Nu_1} \\ \sigma_{u_1u_2} & \sigma_{\theta}^2 + \sigma_{u_2}^2 & \cdots & \sigma_{u_Nu_2} \\ \vdots & \vdots & \ddots & \vdots \\ \sigma_{u_1u_N} & \sigma_{u_2u_N} & \cdots & \sigma_{\theta}^2 + \sigma_{u_N}^2 \end{bmatrix},$$
(A.5)

where $\sigma_{u_i u_j} \equiv C(u_i, u_j)$ (where C stands for covariance) if $i \neq j$ and $\sigma_{u_i}^2 = V(u_i)$ for i = jbecause $C(\lambda_i + u_i, \lambda_j + u_j) = C(u_i, u_i) = V(u_i)$. The next step is to multiply both sides of equation (A.2) by the weighting factor $w \equiv 1/\sqrt{\Omega}$:

$$\mathbf{\Omega}^{-1/2}\hat{\theta} = \mathbf{\Omega}^{-1/2}\mathbf{X}\psi + \mathbf{\Omega}^{-1/2}\mathbf{v},\tag{A.6}$$

where $E(\Omega^{-1/2}\mathbf{v}) = \mathbf{0}$ and $V(\Omega^{-1/2}\mathbf{v}) = \Omega^{-1}V(\mathbf{v}) = \Omega^{-1}\Omega = \mathbf{I}_N$. Applying OLS to (A.6) provides consistent and efficient estimates of ψ . Of course, the above procedure is only operational if all elements of Ω are known or can be estimated.

The third step consists of estimating the elements of Ω . In doing so, we assume that all sources of dependency across observations—except for sample dependency—have been controlled for in the meta-regression. The elements of Ω on the main diagonal of (A.5) can be obtained in the same way as in Section 3.2. The formula differs slightly from that in footnote 21:

$$\hat{\sigma}_{\theta}^{2} = \frac{Q - (N - M - 2)}{c}, \qquad c \equiv \sum_{i=1}^{N} w_{i} - \left[\sum_{i=1}^{N} w_{i}^{2} / \sum_{i=1}^{N} w_{i}\right], \tag{A.7}$$

where $\hat{\sigma}_{u_i}^2 \equiv (\text{se}(\hat{\theta}_i))^2$ is defined as before and w_i are the weights derived from the fixed effects analysis. The off-diagonal elements are in some cases non-zero because there is sample overlap across studies. In formal terms, $\sigma_{u_i u_j} = \rho_{u_i u_j} \sqrt{\sigma_{u_i}^2} \sqrt{\sigma_{u_j}^2}$, where $\rho_{u_i u_j}$ is the correlation coefficient between u_i and u_j . We thus need an estimate of $\rho_{u_i u_j}$. To this end, we define $\hat{\rho}_{u_i u_j} \equiv n_{ij}/(n_i + n_j + n_{ij})$, where n_{ij} is the number of observations in overlapping samples iand j (i.e., the samples that give rise to $\hat{\theta}_i$ and $\hat{\theta}_j$) and n_i (n_j) is the number of observations used to estimate $\hat{\theta}_i$ $(\hat{\theta}_j)$. If two samples do not overlap then $\hat{\rho}_{u_i u_j} = 0$, whereas if they completely overlap then $\hat{\rho}_{u_i u_j} = 1$.

We do not make an attempt to recover estimates for the fixed effects model because: (i) the random effects estimator is statistically preferred; and (ii) the fixed effects would require imposing further restrictions on Ω . The latter argument can be illustrated as follows. If we were to assume a fixed effects model, then σ_{θ}^2 would drop out from the matrix Ω . In that case, any two columns with $\rho_{u_i u_j} = 1$ (which assumes completely overlapping samples), would yield a singular Ω matrix.

1 Ratner (1983) ^a US 0.277 0.099 2 Da Costa, Ellson, and Martin (1987) US 0.281 0.091 3 Aschauer (1989a) US 0.240 0.047 4 Aschauer (1989b) G-7 countries 0.410 0.130 5 Ram and Ramsey (1989) US 0.240 0.055 6 Munnell (1990a) US 0.330 0.066 7 Munnell (1990b) US 0.001 0.031 9 Eisner (1991) US 0.042 0.127 10 Tatom (1991) US 0.042 0.127 12 Garcia-Milà and McGuire (1992) US 0.042 0.127 13 Bajo-Rubio and Sosvilla-Rivero (1993) Spain 0.190 0.007 14 Finn (1993) US 0.040 0.013 15 Mas, Maudos, Pérez, and Uriel (1993) Spain 0.066 0.118 16 Munnell (1993) US -0.062 0.051 17 Prud'Homme (1993) France 0.073	No.	Author(s)	Country	Estimate $\hat{\theta}_i$	$\begin{array}{c} \text{St.Error} \\ \text{se}(\hat{\theta}_i) \end{array}$
2 Da Costa, Ellson, and Martin (1987) US 0.281 0.091 3 Aschauer (1989a) US 0.240 0.047 4 Aschauer (1989b) G-7 countries 0.410 0.130 5 Ram and Ramsey (1989) US 0.240 0.055 6 Munnell (1990a) US 0.330 0.066 7 Munnell (1990b) US 0.081 0.031 9 Eisner (1991) US 0.042 0.127 10 Tatom (1991) US 0.042 0.127 11 Berndt and Hansson (1992) Sweden 0.687 0.220 12 Garcia-Milà and McGuire (1992) US 0.044 0.007 13 Bajo-Rubio and Sosvilla-Rivero (1993) Spain 0.190 0.007 14 Finn (1993) US 0.046 0.013 16 Munnell (1993) US 0.040 0.013 17 Prud'Homme (1993) France 0.073 0.036 18 Eisner (1994) US 0.022	1	Ratner $(1983)^{a}$	US	0.277	0.099
3 Aschauer (1989a) US 0.240 0.047 4 Aschauer (1989b) G-7 countries 0.410 0.130 5 Ram and Ramsey (1989) US 0.330 0.066 6 Munnell (1990a) US 0.330 0.066 7 Munnell (1990b) US 0.081 0.031 9 Eisner (1991) US 0.042 0.127 10 Tatom (1991) US 0.042 0.127 11 Berndt and Hansson (1992) Sweden 0.687 0.220 12 Garcia-Milà and McGuire (1992) US 0.045 0.0051 13 Bajo-Rubio and Sosvilla-Rivero (1993) Spain 0.190 0.0071 14 Finn (1993) US 0.158 0.077 15 Mas, Maudos, Pérez, and Uriel (1993) Spain 0.066 0.118 16 Munnell (1993) US 0.270 0.068 19 Evans and Karras (1994a) US 0.270 0.068 19 Evans and Karras (1994b) 7 OECD countries	2	Da Costa, Ellson, and Martin (1987)	US	0.281	0.091
4 Aschauer (1989b) G-7 countries 0.410 0.130 5 Ram and Ramsey (1989) US 0.240 0.055 6 Munnell (1990a) US 0.330 0.066 7 Munnell (1990b) US 0.081 0.031 9 Eisner (1991) US 0.081 0.031 9 Eisner (1991) US 0.042 0.127 10 Tatom (1991) US 0.045 0.005 13 Bajo-Rubio and Sosvilla-Rivero (1993) Spain 0.190 0.007 14 Finn (1993) US 0.045 0.007 15 Mas, Maudos, Pérez, and Uriel (1993) Spain 0.166 0.118 16 Munnell (1993) US 0.040 0.013 17 Prud'Homme (1993) France 0.073 0.036 18 Eisner (1994) US -0.062 0.051 20 Evans and Karras (1994a) US -0.062 0.051 21 Ferreira (1994) G7 countries -0.175 0.107 21 F	3	Aschauer (1989a)	US	0.240	0.047
5 Ram and Ramsey (1989) US 0.240 0.055 6 Munnell (1990a) US 0.330 0.066 7 Munnell (1990b) US 0.061 0.031 9 Eisner (1991) US 0.081 0.031 9 Eisner (1991) US 0.042 0.127 10 Tatom (1991) US 0.042 0.127 11 Berndt and Hansson (1992) Sweden 0.687 0.220 12 Garcia-Milà and McGuire (1992) US 0.045 0.005 13 Bajo-Rubio and Sosvilla-Rivero (1993) Spain 0.190 0.007 14 Finn (1993) US 0.046 0.138 16 Munnell (1993) Spain 0.190 0.007 14 Finn (1993) France 0.073 0.336 17 Prud'Homme (1993) France 0.073 0.366 18 Eisner (1994) US -0.062 0.151 20 Evans and Karras (1994a) US -0.022 0.131	4	Aschauer (1989b)	G-7 countries	0.410	0.130
6Munnell (1990a)US 0.330 0.066 7Munnell (1990b)US 0.060 0.016 8Duffy-Deno and Eberts (1991)US 0.081 0.031 9Eisner (1991)US 0.042 0.127 10Tatom (1991)US 0.042 0.127 11Berndt and Hansson (1992)Sweden 0.687 0.220 12Garcia-Milà and McGuire (1992)US 0.045 0.005 13Bajo-Rubio and Sosvilla-Rivero (1993)Spain 0.190 0.007 14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US -0.062 0.051 20Evans and Karras (1994a)US -0.062 0.051 21Ferreira (1994)G7 countries -0.175 0.107 21Ferreira (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)US -0.022 0.131 25Ai and Cassou (1995)US 0.174 0.244 26Andrews and Swanson (1995)US 0.174 0.244 26Andrews and Swanson (1995)US 0.110 0.188 27Baltagi and Pinnoi (1995)	5	Ram and Ramsey (1989)	US	0.240	0.055
7Mumell (1990b)US 0.060 0.016 8Duffy-Deno and Eberts (1991)US 0.081 0.031 9Eisner (1991)US 0.042 0.127 10Tatom (1991)US 0.042 0.127 11Berndt and Hansson (1992)Sweden 0.687 0.220 12Garcia-Milà and McGuire (1992)US 0.045 0.005 13Bajo-Rubio and Sosvilla-Rivero (1993)Spain 0.190 0.007 14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.366 18Eisner (1994)US -0.062 0.051 20Evans and Karras (1994a)US -0.062 0.051 21Ferreira (1994)67 countries -0.175 0.107 21Ferreira (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)US 0.174 0.024 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.110 0.182 27Baltagi and Pinnoi (1995)US 0.100 0.275 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Mil	6	Munnell (1990a)	US	0.330	0.066
8 Duffy-Deno and Eberts (1991) US 0.081 0.031 9 Eisner (1991) US 0.077 0.031 10 Tatom (1991) US 0.042 0.127 11 Berndt and Hansson (1992) Sweden 0.687 0.220 12 Garcia-Milà and McGuire (1992) US 0.045 0.005 13 Bajo-Rubio and Sosvilla-Rivero (1993) Spain 0.190 0.007 14 Finn (1993) US 0.158 0.077 15 Mas, Maudos, Pérez, and Uriel (1993) Spain 0.066 0.118 16 Munnell (1993) US 0.040 0.013 17 Prud'Homme (1993) France 0.070 0.068 19 Evans and Karras (1994) US -0.062 0.051 20 Evans and Karras (1994) US -0.022 0.131 21 Ferreira (1994) G7 countries 0.185 0.464 22 Holtz-Eakin (1994) LS	7	Munnell (1990b)	US	0.060	0.016
9Eisner (1991)US 0.077 0.031 10Tatom (1991)US 0.042 0.127 11Berndt and Hansson (1992)Sweden 0.687 0.220 12Garcia-Milà and McGuire (1992)US 0.045 0.005 13Bajo-Rubio and Sosvilla-Rivero (1993)Spain 0.190 0.007 14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.033 17Prud'Homme (1993)France 0.073 0.066 18Eisner (1994)US 0.270 0.068 19Evans and Karras (1994a)US -0.062 0.511 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.110 0.018 26Andrews and Swanson (1995)US 0.002 0.100 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995a)US 0.012 0.064 34Holtz-Eakin and Schwartz (1995a)US -0.070 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 $0.$	8	Duffy-Deno and Eberts (1991)	US	0.081	0.031
10Tatom (1991)US 0.042 0.127 11Berndt and Hansson (1992)Sweden 0.687 0.220 12Garcia-Milà and McGuire (1992)US 0.045 0.005 13Bajo-Rubio and Sosvilla-Rivero (1993)Spain 0.190 0.007 14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US -0.062 0.051 20Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.004 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US -0.007 0.225 31Sturm and De Haan (1995)US -0.007 0.225 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.775 33Holtz-Eakin and Schwartz (1995b)US <td>9</td> <td>Eisner (1991)</td> <td>US</td> <td>0.077</td> <td>0.031</td>	9	Eisner (1991)	US	0.077	0.031
11Berndt and Hansson (1992)Sweden 0.687 0.220 12Garcia-Milà and McGuire (1992)US 0.045 0.005 13Bajo-Rubio and Sosvilla-Rivero (1993)Spain 0.190 0.007 14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.366 18Eisner (1994)US 0.220 0.068 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994a)US -0.062 0.051 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.100 0.84 27Baltagi and Pinnoi (1995)US 0.112 0.064 29Holtz-Eakin and Schwartz (1995a)US -0.078 0.275 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 31Sturm and De Haan (1995)The Neth	10	Tatom (1991)	US	0.042	0.127
12Garcia-Milà and McGuire (1992)US 0.045 0.005 13Bajo-Rubio and Sosvilla-Rivero (1993)Spain 0.190 0.007 14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US 0.270 0.068 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.044 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)US 0.174 0.024 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.110 0.188 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995a)US 0.002 0.002 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Schwartz (1995b)US -0.058 0.075 34Holtz-Eakin and Lovely (1996) </td <td>11</td> <td>Berndt and Hansson (1992)</td> <td>Sweden</td> <td>0.687</td> <td>0.220</td>	11	Berndt and Hansson (1992)	Sweden	0.687	0.220
13Bajo-Rubio and Sosvilla-Rivero (1993)Spain 0.190 0.007 14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US 0.270 0.068 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995a)US 0.012 0.040 30Holtz-Eakin and Schwartz (1995a)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.775 32Garcia-Milà, McGuire, and Porter (1996)US -0.044 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.225 36Otto and Voss (1996)US	12	Garcia-Milà and McGuire (1992)	US	0.045	0.005
14Finn (1993)US 0.158 0.077 15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US 0.270 0.068 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.110 0.018 26Andrews and Swanson (1995)US 0.002 0.100 27Baltagi and Pinnoi (1995)US 0.012 0.064 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US -0.007 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.286 36Otto and Voss (1996)Spain 0.286	13	Bajo-Rubio and Sosvilla-Rivero (1993)	Spain	0.190	0.007
15Mas, Maudos, Pérez, and Uriel (1993)Spain 0.066 0.118 16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US 0.270 0.668 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.110 0.018 26Andrews and Swanson (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.046 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.044 0.086 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) <td>14</td> <td>Finn (1993)</td> <td>US</td> <td>0.158</td> <td>0.077</td>	14	Finn (1993)	US	0.158	0.077
16Munnell (1993)US 0.040 0.013 17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US 0.270 0.068 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.002 0.100 27Baltagi and Pinnoi (1995)US 0.012 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.064 29Holtz-Eakin and Schwartz (1995b)US -0.007 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.880 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.803 37Ram (1996)Spain<	15	Mas, Maudos, Pérez, and Uriel (1993)	Spain	0.066	0.118
17Prud'Homme (1993)France 0.073 0.036 18Eisner (1994)US 0.270 0.068 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 $OECD$ countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.046 29Holtz-Eakin and Schwartz (1995b)US -0.007 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.086 34Khanam (1996)Canada 0.140 0.886 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)S3 LDCs 0.135 0.047 38Crowder and	16	Munnell (1993)	ŪS	0.040	0.013
18Eisner (1994)US 0.270 0.068 19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.110 0.018 26Andrews and Swanson (1995)US 0.100 0.002 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 44Khanam (1996)Canada 0.140 0.800 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.800 37Ram (1996)S3 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045	17	Prud'Homme (1993)	France	0.073	0.036
19Evans and Karras (1994a)US -0.062 0.051 20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.100 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)Sa LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045	18	Eisner (1994)	US	0.270	0.068
20Evans and Karras (1994b)7 OECD countries -0.175 0.107 21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)S3 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045	19	Evans and Karras (1994a)	US	-0.062	0.051
21Ferreira (1994)67 countries 0.185 0.046 22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)S3 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045	20	Evans and Karras (1994b)	7 OECD countries	-0.175	0.107
22Holtz-Eakin (1994)US -0.022 0.131 23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.255 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.800 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kavanagh (1997)Us 0.144 0.298	21	Ferreira (1994)	67 countries	0.185	0.046
23Mas, Maudos, Pérez, and Uriel (1994)Spain 0.230 0.084 24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.225 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.80 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)S3 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.044 39Kavanach (1997)Ireland 0.144 0.298	22	Holtz-Eakin (1994)	US	-0.022	0.131
24Otto and Voss (1994)Australia 0.381 0.143 25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.800 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) 53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kavanagh (1997)Ireland 0.144 0.298	23	Mas, Maudos, Pérez, and Uriel (1994)	Spain	0.230	0.084
25Ai and Cassou (1995)US 0.174 0.024 26Andrews and Swanson (1995)US 0.110 0.018 27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.800 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) 53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kavanagh (1997)Us 0.144 0.298	24	Otto and Voss (1994)	Australia	0.381	0.143
26Andrews and Swanson (1995)US0.1100.018 27 Baltagi and Pinnoi (1995)US0.0020.100 28 De la Fuentes and Vives (1995)Spain0.2120.064 29 Holtz-Eakin and Schwartz (1995a)US0.1120.040 30 Holtz-Eakin and Schwartz (1995b)US -0.007 0.025 31 Sturm and De Haan (1995)The Netherlands0.7800.275 32 Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33 Holtz-Eakin and Lovely (1996)US -0.144 0.064 34 Khanam (1996)Canada0.1400.080 35 Mas, Maudos, Pérez, and Uriel (1996)Spain0.0860.025 36 Otto and Voss (1996)53 LDCs0.1350.047 38 Crowder and Himarios (1997)US0.2940.045 39 Kavanagh (1997)Us0.1440.208	25	Ai and Cassou (1995)	US	0.174	0.024
27Baltagi and Pinnoi (1995)US 0.002 0.100 28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.80 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) 53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kayanagh (1997)Iraland 0.144 0.298	26	Andrews and Swanson (1995)	US	0.110	0.018
28De la Fuentes and Vives (1995)Spain 0.212 0.064 29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kayanagh (1997)Ireland 0.144 0.298	27	Baltagi and Pinnoi (1995)	US	0.002	0.100
29Holtz-Eakin and Schwartz (1995a)US 0.112 0.040 30Holtz-Eakin and Schwartz (1995b)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) 53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kayanagh (1997)Ireland 0.144 0.298	28	De la Fuentes and Vives (1995)	Spain	0.212	0.064
30Holtz-Eakin and Schwartz (1995b)US -0.007 0.025 31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) 53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kayanagh (1997)Ireland 0.144 0.298	29	Holtz-Eakin and Schwartz (1995a)	ŪS	0.112	0.040
31Sturm and De Haan (1995)The Netherlands 0.780 0.275 32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996)53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kayanagh (1997)Ireland 0.144 0.298	30	Holtz-Eakin and Schwartz (1995b)	US	-0.007	0.025
32Garcia-Milà, McGuire, and Porter (1996)US -0.058 0.075 33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) 53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kayanagh (1997)Ireland 0.144 0.298	31	Sturm and De Haan (1995)	The Netherlands	0.780	0.275
33Holtz-Eakin and Lovely (1996)US -0.144 0.064 34Khanam (1996)Canada 0.140 0.080 35Mas, Maudos, Pérez, and Uriel (1996)Spain 0.086 0.025 36Otto and Voss (1996)Australia 0.168 0.080 37Ram (1996) 53 LDCs 0.135 0.047 38Crowder and Himarios (1997)US 0.294 0.045 39Kayanagh (1997)Ireland 0.144 0.298	32	Garcia-Milà, McGuire, and Porter (1996)	US	-0.058	0.075
34 Khanam (1996) Canada 0.140 0.080 35 Mas, Maudos, Pérez, and Uriel (1996) Spain 0.086 0.025 36 Otto and Voss (1996) Australia 0.168 0.080 37 Ram (1996) 53 LDCs 0.135 0.047 38 Crowder and Himarios (1997) US 0.294 0.045 39 Kayanagh (1997) Ireland 0.144 0.298	33	Holtz-Eakin and Lovely (1996)	US	-0.144	0.064
35 Mas, Maudos, Pérez, and Uriel (1996) Spain 0.086 0.025 36 Otto and Voss (1996) Australia 0.168 0.080 37 Ram (1996) 53 LDCs 0.135 0.047 38 Crowder and Himarios (1997) US 0.294 0.045 39 Kayanagh (1997) Ireland 0.144 0.298	34	Khanam (1996)	Canada	0.140	0.080
36 Otto and Voss (1996) Australia 0.168 0.080 37 Ram (1996) 53 LDCs 0.135 0.047 38 Crowder and Himarios (1997) US 0.294 0.045 39 Kayanagh (1997) Ireland 0.144 0.298	35	Mas, Maudos, Pérez, and Uriel (1996)	Spain	0.086	0.025
37 Ram (1996) 53 LDCs 0.135 0.047 38 Crowder and Himarios (1997) US 0.294 0.045 39 Kayanagh (1997) Ireland 0.144 0.298	36	Otto and Voss (1996)	Australia	0.168	0.080
38 Crowder and Himarios (1997) US 0.294 0.045 39 Kayanagh (1997) Ireland 0.144 0.208	37	Ram (1996)	53 LDCs	0.135	0.047
30 Kayanagh (1997)	38	Crowder and Himarios (1997)	US	0.294	0.045
0.144 0.290	39	Kavanagh (1997)	Ireland	0.144	0.298

Table A1. Studies Included in the Meta-Data Set

-

(Continued on next page)

Table A1 (Continued)

No.	Author(s)	Country	Estimate $\hat{\theta}_i$	$\begin{array}{c} \text{St.Error} \\ \text{se}(\hat{\theta}_i) \end{array}$
40	Kelejian and Robinson (1997)	US	-0.023	0.064
41	Khan and Kumar (1997)	95 LDCs	0.290	0.090
42	Nazmi and Ramirez (1997)	Mexico	0.129	0.028
43	Vijverberg, Vijverberg, and Gamble (1997)	US	0.481	0.111
44	Boarnet (1998)	US	0.257	0.052
45	Erenburg (1998)	US	0.290	0.070
46	Flores de Frutos, Diez, and Amaral (1998)	Spain	0.210	0.070
47	Moreno, Artís, López-Bazo and Suriñach (1998)	Spain	0.049	0.021
48	Nourzad (1998)	ŪS	0.340	0.200
49	Otto and Voss (1998)	Australia	0.059	0.004
50	Ramirez (1998)	Mexico	0.590	0.298
51	Cadot, Roller, and Stephan (1999)	France	0.085	0.040
52	Delorme, Thompson, and Warren (1999)	US	0.213	0.132
53	Picci (1999)	Italy	0.501	0.032
54	Bonaglia, La Ferrara, and Marvelino (2000)	Italy	0.305	0.043
55	Charlot and Schmitt (2000)	France	0.317	0.021
56	Dessus and Herrera (2000)	28 LDCs	0.130	0.065
57	La Ferrara and Marcelino (2000)	Italy	-0.148	0.048
58	Nourzad (2000)	24 countries	0.529	0.107
59	Yamano and Ohkawara (2000)	Japan	0.148	0.016
60	Yamarik (2000)	US	0.088	0.032
61	Alonso-Carrera and Freire-Séren (2001)	Spain	0.126	0.027
62	Owyong and Thangavelu (2001)	Canada	0.917	0.194
63	Shioji (2001)	Japan	0.241	0.062
64	Stephan (2001)	Germany and France	0.112	0.040
65	Kemmerling and Stephan (2002)	Germany	0.170	0.037
66	Ligthart (2002)	Portugal	0.370	0.029
67	Rubio, Roldán, and Garcés (2002)	Spain	0.040	0.018
68	Stephan (2003)	Germany	0.537	0.134
69	Rodríguez-Valez and Yarias Sampedro (2004)	Spain	0.160	0.029
70	Cantos, Gumbau, and Maudos (2005)	Spain	0.042	0.016
71	Kataoka (2005)	Japan	0.185	0.018
72	Kawaguchi, Ohtake, and Tamada (2005)	Japan	0.180	0.160
73	Le and Suruga (2005)	105 countries	0.076	0.024
74	Berechman, Ozmen, and Ozbay (2006)	US	0.035	0.019
75	Creel and Poilon (2006)	6 EMU countries	0.140	0.012
76	Kamps (2006)	22 OECD countries	0.221	0.060
	Average		0.193	0.198

^a The estimates considered are those replicated by Tatom (1991) using revised data for the same period. Ratner's (1983) original estimate amounts to 0.056.

References

- AARON, H. J. (1990): "Discussion," in *Is There a Shortfall in Public Capital Investment?*, ed. by A. H. Munnell. Federal Reserve Bank of Boston, Boston.
- AI, C., AND S. P. CASSOU (1995): "A Normative Analysis of Public Capital," Applied Economics, 27, 1201–1209.
- ALONSO-CARRERA, J., AND M. J. FREIRE-SERÉN (2001): "Infrastructuras Públicas y Desarrollo Económico de Galicia," mimeo, Universidade de Vigo, Vigo.
- ANDREWS, K., AND J. SWANSON (1995): "Does Public Infrastructure Affect Regional Performance?," Growth and Change, 26, 204–216.
- ASCHAUER, D. A. (1989a): "Is Public Expenditure Productive?," Journal of Monetary Economics, 23, 177–200.
 - (1989b): "Public Investment and Productivity Growth in the Group of Seven," *Economic Perspectives*, 13, 17–25.
 - (1990): "Why is Public Capital Important?," in *Is There a Shortfall in Public Capital Investment?*, ed. by A. H. Munnell. Federal Reserve Bank of Boston, Boston.
- BAJO-RUBIO, O., AND S. SOSVILLA-RIVERO (1993): "Does Public Capital Affect Private Sector Performance? An Analysis of the Spanish Case," *Economic Modelling*, 10, 179–184.
- BALTAGI, B. D., AND N. PINNOI (1995): "Public Capital Stock and State Productivity Growth: Further Evidence from an Error Components Model," *Empirical Economics*, 20, 351–359.
- BATINA, R. G. (1998): "On the Long Run Effects of Public Capital and Disaggregated Public Capital on Aggregate Output," *International Tax and Public Finance*, 94, 263–281.
- BEGG, C. B. (1994): "Publication Bias," in *The Handbook of Research Synthesis*, ed. by H. Cooper, and L. V. Hedges. Russel Sage Foundation, New York.
- BERECHMAN, J., D. OZMEN, AND K. OZBAY (2006): "Empirical Analysis of Transportation Investment and Economic Development at State, County, and Municipality Levels," *Transportation*, 33, 537–551.
- BERNDT, E. R., AND B. HANSSON (1992): "Measuring the Contribution of Public Infrastructure Capital in Sweden," *Scandinavian Journal of Economics*, 94, S151–S161.
- BIJMOLT, T. H. A., AND R. G. M. PIETERS (2001): "Meta-Analysis in Marketing when Studies Contain Multiple Measurements," *Marketing Letters*, 12, 157–169.
- BOARNET, M. G. (1998): "Spillovers and the Locational Effects of Public Infrastructure," Journal of Regional Science, 38, 381–400.
- BONAGLIA, F., E. LA FERRARA, AND M. MARCELINO (2000): "Public Capital and Economic Performance: Evidence from Italy," Working Papers No. 163, IGIER, Bocconi University.

- BUTTON, K. (1998): "Infrastructure Investment, Endogenous Growth, and Economic Convergence," *The Annals of Regional Science*, 32, 145–162.
- BUTTON, K., AND T. WEYMAN-JONES (1992): "Ownership Structure, Institutional Organization, and Measured Inefficiency," *American Economic Review*, 82, 439–445.
- BUTTON, K. J., AND P. RIETVELD (2000): "A Meta Analysis of the Impact of Infrastructure Policy on Regional Development," in *Regional Cohesion and Competition in the Age of Globalization*, ed. by H. Kohno, P. Nijkamp, and J. Poot. Edward Elgar, Boston.
- CADOT, O., L. ROLLER, AND A. STEPHAN (1999): "A Political Economy Model of Infrastructure Allocation: An Empirical Assessment," Working Paper No. 2336, Centre for Economic and Policy Research, London.
- CAMERON, A. C., J. B. GELBACH, AND D. L. MILLER (2007): "Bootstrap-Based Improvements for Inference with Clustered Errors," *Review of Economics and Statitics*, forthcoming.
- CANTOS, P., M. GUMBAU, AND J. MAUDOS (2005): "Transport Infrastructures, Spillover Effects and Regional Growth: Evidence of the Spanish Case," *Transport Reviews*, 25, 25–50.
- CARD, D., AND A. B. KRUEGER (1995): "Time-Series Minimum-Wage Studies: A Meta-Analysis," American Economic Review, 85, 238–243.
- CHARLOT, S., AND B. SCHMITT (2000): "Public Infrastructure and Economic Growth in France's Regions," Working Paper, UMR INRA ENESAD, Dijon, France.
- CREEL, J., AND G. POILON (2006): "Is Public Capital Productive in Europe?," Documents de Travail de l'OFCE 2006-10, Observatoire Francais des Conjonctures Economiques (OFCE).
- CROWDER, W. J., AND D. HIMARIOS (1997): "Balanced Growth and Public Capital: An Empirical Analysis," *Applied Economics*, 29, 1045–1053.
- DA COSTA, S. J., R. W. ELLSON, AND R. C. MARTIN (1987): "Public Capital, Regional Output, and Developments: Some Empirical Evidence," *Journal of Regional Science*, 27, 419–37.
- DAMALGAS, B. (1995): "A Reconsideration of the Public Sector's Contribution to Growth," Empirical Economics, 20, 385–414.
- DE LA FUENTE, A., AND X. VIVES (1995): "Infrastructure and Education as Instruments of Regional Policy: Evidence from Spain," *Regional Policy and Spain*, 20, 11–51.
- DE MOOIJ, R. A., AND S. EDERVEEN (2003): "Taxation and Foreign Direct Investment: A Synthesis," *International Tax and Public Finance*, 10, 673–693.
- DELORME, C. D., H. G. THOMPSON, AND R. S. WARREN (1999): "Public Infrastructure and Private Productivity: A Stochastic Frontier Approach," *Journal of Macroeconomics*, 2, 563–576.
- DESSUS, S., AND R. HERRERA (2000): "Public Capital and Growth Revisited: A Panel Data Assessment," *Economic Development and Cultural Change*, 48, 407–418.

- DUFFY-DENO, K. T., AND R. W. EBERTS (1991): "Public Infrastructure and Regional Economic Development: A Simultaneous Equations Approach," *Journal of Urban Economics*, 30, 329–343.
- DURLAUF, S., P. JOHNSON, AND J. TEMPLE (2005): "Growth Econometrics," in *Handbook* of *Economic Growth*, Vol. 1 Part 1, ed. by P. Aghion, and S. Durlauf, pp. 555–677. Elsevier, North-Holland, Amsterdam.
- EISNER, R. (1991): "Infrastructure and Regional Economic Performance: Comment," New England Economic Review, September/October, 47–58.
- (1994): "Real Government Saving and the Future," Journal of Economic Behavior and Organization, 23, 111–126.
- ENGLE, R. F., AND C. W. J. GRANGER (1987): "Cointegration and Error Correction: Representation, Estimation, and Testing," *Econometrica*, 55, 251–276.
- ERENBURG, S. J. (1998): "Productivity, Private and Public Capital, and Real Wage in the US," Applied Economics Letters, 5, 491–495.
- EVANS, P., AND G. KARRAS (1994a): "Are Government Activities Productive? Evidence from a Panel of US States," *Review of Economics and Statistics*, 76, 1–11.
- (1994b): "Is Government Capital Productive? Evidence from a Panel of Seven Countries," *Journal of Macroeconomics*, 16, 271–279.
- FERREIRA, P. C. (1994): "The Impact of Public Capital and Public Investment on Economic Growth," Epge ensaios econômicos no. 228, Fundação Getulio Vargas.
- FINN, M. (1993): "Is all Government Capital Productive?," Federal Reserve Bank of Richmond Economic Quarterly, 79, 53–80.
- FLORES DE FRUTOS, R., M. G. DIEZ, AND T. P. AMARAL (1998): "Public Capital Stock and Economic Growth: An Analysis of the Spanish Economy," *Applied Economics*, 322, 985–994.
- GARCIA-MILÀ, T., AND T. J. MCGUIRE (1992): "The Contribution of Publicly Provided Inputs to States' Economies," *Regional Science and Urban Economics*, 22, 229–241.
- GARCIA-MILÀ, T., T. J. MCGUIRE, AND R. H. PORTER (1996): "The Effects of Public Capital in State Level Production Functions Reconsidered," *Review of Economics and Statistics*, 78, 177–180.
- GRAMLICH, E. M. (1994): "Infrastructure Investment: A Review Essay," Journal of Economic Literature, 32, 1176–1196.
- HEDGES, L. V. (1994): "Fixed Effects Models," in *The Handbook of Research Synthesis*, ed. by H. Cooper, and L. V. Hedges. Russel Sage Foundation, New York.
- HOLTZ-EAKIN, D. (1994): "Public-Sector Capital and the Productivity Puzzle," *Review of Economics and Statistics*, 76, 12–21.

- HOLTZ-EAKIN, D., AND M. E. LOVELY (1996): "Scale Economies, Returns to Variety, and the Productivity of Public Infrastructure," *Regional Science and Urban Economics*, 26, 105–123.
- HOLTZ-EAKIN, D., AND A. E. SCHWARTZ (1995a): "Infrastructure in a Structural Model of Economic Growth," *Regional Science and Urban Economics*, 25, 131–151.

(1995b): "Spatial Productivity Spillovers from Public Infrastructure: Evidence from State Highways," International Tax and Public Finance, 2, 459–468.

- HULTEN, C. R., AND R. M. SCHWAB (1991a): "Is There Too Little Public Capital: Infrastructure and Economic Growth," Conference paper, American Enterprise Institute, Washington, DC.
- (1991b): "Public Capital Formation and the Growth of the Regional Manufacturing Industries," *National Tax Journal*, 40, 121–134.
- IMF (2004): "Public Investment and Fiscal Policy," Staff Memorandum No. SM/04/93, International Monetary Fund, Washington, DC.
- JOHANSEN, S. (1988): "Statistical Analysis of Cointegration Vectors," Journal of Economic Dynamics and Control, 12, 231–254.
- KAMPS, C. (2006): "New Estimates of Government Net Capital Stocks for 22 OECD Countries 1960-2001," *IMF Staff Papers*, 53, 120–150.
- KATAOKA, M. (2005): "Effect of Public Investment on the Regional Economies in Postwar Japan," *Review of Urban and Regional Development Studies*, 17, 115–139.
- KAVANAGH, C. (1997): "Public Capital and Private Sector Productivity in Ireland," Journal of Economic Studies, 24, 72–94.
- KAWAGUCHI, D., F. OHTAKE, AND K. TAMADA (2005): "The Productivity of Public Capital: Evidence from the 1994 Electoral Reform of Japan," Discussion Paper 627, The Institute of Social Economic Research.
- KELEJIAN, H. H., AND D. P. ROBINSON (1997): "Infrastructure Productivity Estimation and its Underlying Econometric Specifications: A Sensitivity Analysis," *Papers in Regional Science*, 76, 115–131.
- KEMMERLING, A., AND A. STEPHAN (2002): "The Contribution of Local Infrastructure to Private Productivity and its Political Economy: Evidence From a Panel of Large German Cities," *Public Choice*, 113, 403–422.
- KHAN, M. S., AND M. S. KUMAR (1997): "Public and Private Investment and the Growth Process in Developing Countries," *Logistics and Transportation Review*, 59, 69–88.
- KHANAM, B. R. (1996): "Highway Infrastructure Capital and Productivity Growth: Evidence from the Canadian Goods-Producing Sector," *Logistics and Transportation Review*, 32, 251–268.
- KODRZYCKI, Y., AND P. YU (2006): "New Approaches to Ranking Economics Journals," Contributions to Economic Analysis and Policy, 5, 1–42.

- LA FERRARA, E., AND M. MARCELLINO (2000): "TFP, Costs, and Public Infrastructure: An Equivocal Relationship," Working Paper No. 176, IGIER, Milan.
- LE, M. V., AND T. SURUGA (2005): "Foreign Direct Investment, Public Expenditure and Economic Growth: The Empirical Evidence for the Period 1970-2001," Applied Economics Letters, 12, 45–49.
- LIGTHART, J. E. (2002): "Public Capital and Output Growth in Portugal: An Empirical Analysis," *European Review of Economics and Finance*, 1, 3–30.
- MAS, M., J. MAUDOS, F. PÉREZ, AND E. URIEL (1993): "Competitividad, Productividad Industrial y Dotaciones de Capital Publico," *Papeles de Economia Española*, 56, 144–160.
- (1994): "Capital Público y Productividad en las Regiones Españolas," Moneda y Credito, 198, 163–192.
- (1996): "Infrastructure and Productivity in the Spanish Regions," *Regional Studies*, 30, 641–50.
- MCMILLIN, W., AND D. SMITH (1994): "A Multivariate Time Series Analysis of the United States Aggregate Production Function," *Empirical Economics*, 19, 659673.
- MERA, K. (1973): "Regional Production Functions and Social Overhead: An Analysis of the Japanese Case," *Regional and Urban Economics*, 3, 157–186.
- MERRIMAN, D. (1990): "Public Capital and Regional Output: Another Look at Some Japanese and American Data," *Regional Science and Urban Economics*, 20, 437–458.
- MIKELBANK, B. A., AND R. W. JACKSON (2000): "The Role of Space in Public Capital Research," *International Regional Science Review*, 23, 235–258.
- MORENO, R., M. ARTÍS, E. LÓPEZ-BAZO, AND J. SURIÑACH (1998): "Evidence on the Complex Link Between Infrastructure and Regional Growth," Working papers in economics no. 19, Universitat de Barcelona. Espai de Recerca en Economia, Barcelona.
- MUNNELL, A. H. (1990a): "Why Has Productivity Growth Declined? Productivity and Public Investment," New England Economic Review, January/February, 2–22.
 - (1990b): "How Does Public Infrastructure Affect Regional Economic Performance?," *New England Economic Review*, September/October, 11–32.

(1991): "Is There a Shortfall in Public Capital Investment? An Overview," New England Economic Review, May/June, 2–35.

(1992): "Policy Watch: Infrastructure Investment and Economic Growth," *Journal of Economic Perspectives*, 6, 189–198.

(1993): "An Assessment of Trends in and Economic Impacts of Infrastructure Investment," in *Infrastructure Policies for the 1990s.* OECD, Paris.

NAZMI, N., AND M. D. RAMIREZ (1997): "Public and Private Investment and Economic Growth in Mexico," *Contemporary Economic Policy*, 15, 65–75.

NOURZAD, F. (1998): "Infrastructure Capital and Private Sector Productivity: A Dynamic Analysis," *Quarterly Journal of Business and Economics*, 37, 13–25.

(2000): "The Productivity Effect of Government Capital in Developing and Industrialized Countries," *Applied Economics*, 32, 1181–1187.

- OTTO, G. D., AND G. M. VOSS (1994): "Public Capital and Private Sector Productivity," *Economic Record*, 70, 121–132.
 - (1996): "Public Capital and Private Production in Australia," *Southern Economic Journal*, 62, 723–738.
- (1998): "Is Public Capital Provision Efficient?," Journal of Monetary Economics, 42, 47–66.
- OYWONG, D. T., AND S. M. THANGAVELU (2001): "An Empirical Study on Public Capital Spillovers From the USA to Canada," *Journal of Monetary Economics*, 33, 1493–1499.
- PEREIRA, A. M., AND O. ROCA SAGALES (1999): "Public Capital Formation and Regional Development in Spain," *Review of Development Economics*, 3, 281–294.

(2003): "On the Regional Impact of Public Capital Formation in Spain," Working Paper No. 03.05, Universidad Autonoma de Barcelona, Barcelona.

- PFAHLER, W., U. HOFMANN, AND W. BONTE (1996): "Does Extra Pubic Infrastructure Capital Matter?," *FinanzArchiv*, 53, 68–112.
- PICCI, L. (1999): "Productivity and Infrastructure in the Italian Regions," *Giornale degli* Economisti e Annali di Economia, 58, 329–353.
- PINNOI, N. (1994): "Public Infrastructure and Private Production: Measuring Relative Contributions," Journal of Economic Behavior and Organization, 23, 127–148.
- PRUD'HOMME, R. (1993): "Assessing the Role of Infrastructure in France by Means of Regionally Estimated Production Functions," Discussion paper, Observatoire de L'Economie et de las Instituitions Locales, Paris.
- RAM, R. (1996): "Productivity of Public Capital and Private Investment in Developing Countries: A Broad International Perspective," World Development, 24, 1373–1378.
- RAM, R., AND D. D. RASMEY (1989): "Government Capital and Private Output in the United States: Additional Evidence," *Economics Letters*, 30, 223–226.
- RAMIREZ, M. D. (1998): "Does Public Investment Enhance Productivity Growth in Mexico? A Cointegration Analysis," *Eastern Economic Journal*, 24, 63–82.
- RATNER, J. B. (1983): "Government Capital and the Production Function for US Private Output," *Economics Letters*, 13, 213–217.
- RODRÍGUEZ-VÁLEZ, J., AND C. YARIAS SAMPEDRO (2004): "Desbordamiento Espacial de la Productividad de las Infraestructuras: Una Aplicación con Fronteras Estocásticas," *Estudios de Economía Aplicada*, 22, 1–16.

- ROMP, W., AND J. DE HAAN (2007): "Public Capital and Economic Growth: A Critical Survey," *Perspektiven der Wirtschaftspolitik*, 8, 6–52.
- ROSE, A. K., AND T. D. STANLEY (2005): "A Meta-Analysis of the Effect of Common Currencies on International Trade," *Journal of Economic Perspectives*, 19, 347–365.
- ROSENTHAL, R. (1991): Meta-Analytic Procedures for Social Research. Sage, Newbury Park.
- RUBIO, O. B., C. D. ROLDÁN, AND M. D. M. GARCÉS (2002): "Optimal Endowments of Public Investment: An Empirical Analysis for the Spanish Regions," Economic Working Papers at Centro de Estudios Andaluces E2002/14, Centro de Estudios Andaluces.
- SHADISH, W. R., AND C. K. HADDOCK (1994): "Fixed Effects Models," in *The Handbook* of *Research Synthesis*, ed. by H. Cooper, and L. V. Hedges. Russel Sage Foundation, New York.
- SHIOJI, E. (2001): "Public Capital and Economic Growth: A Convergence Approach," Journal of Economic Growth, 6, 205–227.
- STANLEY, T. D. (1998): "New Wine in Old Bottles: A Meta-Analysis of Ricardian Equivalence," Southern Economic Journal, 64, 713–727.
- ——— (2001): "Wheat From Chaff: Meta-Analysis as Quantitative Literature Review," Journal of Economic Perspectives, 15, 131–150.
 - (2005): "Beyond Publication Bias," Journal of Economic Surveys, 19, 309–345.
- STANLEY, T. D., AND S. B. JARRELL (1989): "Meta-Regression Analysis: A Quantitative Method of Literature Reviews," *Journal of Economic Surveys*, 3, 161–170.
- STEPHAN, A. (2001): "Regional Infrastructure Policy and its Impact on Productivity: A Comparison of Germany and France," DIW Working Paper No. 6748, DIW, Berlin.
- (2003): "Assessing the Contribution of Public Capital to Private Production: Evidence from the German Manufacturing Sector," *International Review of Applied Economics*, 17, 399–418.
- STURM, J.-E., AND J. DE HAAN (1995): "Is Public Expenditure Really Productive: New Evidence for the USA and the Netherlands," *Economic Modelling*, 12, 60–72.
- STURM, J.-E., J. JACOBS, AND P. GROOTE (1999): "Output Effects of Infrastructure Investment in the Netherlands, 18531913," *Journal of Macroeconomics*, 21, 355380.
- STURM, J.-E., G. H. KUPER, AND J. DE HAAN (1998): "Modelling Government Investment and Economic Growth on a Macro Level: A Review," in *Market Behaviour and Macroeconomic Modelling*. MacMillan Press Ltd, London.
- TATOM, J. A. (1991): "Public Capital and Private Sector Performance," *Federal Reserve* Bank of St. Louis Review, 73, 3–15.
- VAN DER SLUIS, VAN PRAAG, M., AND W. VIJVERBERG (2005): "Entrepreneurship Selection and Performance: A Meta-Analysis of the Impact of Education in Less Developed Countries," World Bank Economic Review, 19, 225–261.

- VIJVERBERG, W. P. M., C.-P. VIJVERBERG, AND J. L. GAMBLE (1997): "Public Capital and Private Productivity," *Review of Economics and Statistics*, 79, 267–278.
- WHITE, W. D. (1994): "Scientific Communication and Literature Retrieval," in *The Handbook of Research Synthesis*, ed. by H. Cooper, and L. V. Hedges. Russel Sage Foundation, New York.
- YAMANO, N., AND T. OHKAWARA (2000): "The Regional Allocation of Public Investment: Efficiency or Equity?," *Journal of Regional Science*, 40, 205–229.
- YAMARIK, S. (2000): "The Effect of Public Infrastructure on Private Production During 1977-96," mimeo, University of Akron, Akron, Ohio.